Contracting, Exclusivity and the Formation of Supply Networks with Downstream Competition

Paolo Ramezzana*

October 22, 2017

Abstract

This paper studies the endogenous formation of supply networks in bilateral oligopoly by analyzing a model of contracting with transfers in which each firm optimizes its entire set of contracts across multiple bilateral negotiations. Because of downstream competition, industry profits are not necessarily maximized when all supply links are active and the supply networks that constitute coalition-proof Nash equilibria of the contracting game may differ from those that maximize industry profits. I first demonstrate that, in the absence of public commitment, all marginal input prices in any self-enforcing supply network are equal to the marginal cost of production. I then explore how a number of factors – such as supplier and retailer differentiation and the availability of exclusive contracts – affect the structure of equilibrium supply networks, profits and welfare. I also explore how the analysis changes if firms cannot use transfers or long-term contracts at the network-formation stage and must instead engage in ex-post bilateral bargaining with the associated hold-up problems.

Keywords: Bilateral oligopoly, contracting with transfers, exclusive contracts, supply network formation.

JEL Classification: D43, D85, L13, L14.

^{*}U.S. Federal Trade Commission, 600 Pennsylvania Avenue N.W., Washington, DC 20580; pramezzana@ftc.com. I would like to thank Jonathan Baker, Bhaskar Dutta, Dan O'Brien, David Schmidt, Marius Schwartz, Michael Vita and Alison Weingarden for helpful comments on this or previous versions. Any opinions expressed in this paper are mine alone and do not represent those of the Federal Trade Commission or its Commissioners.

1 Introduction

In many industries a number of differentiated but competing upstream firms ("suppliers") distribute their products through a number of differentiated but competing downstream firms ("retailers"). In some of these industries most retailers distribute the products of most suppliers, i.e., most potential supply links are active. In other industries, instead, each retailer distributes the products of a different supplier and some important potential supply links remain inactive. Examples of the latter include the exclusive distribution of many sports events and films in the pay TV industry – see, e.g., OECD (2013) and Weeds (2016) – and, until a few years ago, of the iPhone and other models of smartphones in the wireless telecommunication industry – see, e.g., Sinkinson (2014). Moreover, in some industries the structure of supply networks appears to be changing over time. For example, the wireless telecommunication industry has recently moved away from smartphone exclusivity, whereas the healthcare industry appears to be moving in the opposite direction, with an increasing number of health insurance companies offering networks that restrict the ability of patients to choose among different healthcare providers (see, e.g., Ho and Lee, 2017).

This paper presents a model of contracting and competition that can shed some light on the endogenous emergence of different structures of supply networks in markets like the ones discussed above. In this model, first all suppliers and retailers simultaneously have an opportunity to negotiate nonlinear vertical contracts, then all retailers with at least one contract compete in the downstream market. Importantly, when negotiating its contracts in the first stage, each firm makes use of all the information at its disposal and optimizes its strategy across multiple bilateral negotiations. The model is used to study the effects of nonexclusive and exclusive contracts on equilibrium supply networks, profits and welfare in environments with varying degrees of supplier and retailer differentiation.

Bilateral trade and contracting in settings with multiple firms on both sides of a market have been studied by at least two streams of literature. A substantial theoretical literature has studied trade in buyer-seller networks (e.g., Kranton and Minehart, 2001; Ostrovsky, 2008; and Elliott, 2015) and contracting with multiple principals and agents (e.g., Prat and Rustichini, 2003). However, this literature has focused exclusively on the case in which buyers do not compete in the downstream market. Downstream competition, and specifically the ability of supply contracts to affect such competition, plays instead a crucial role in this paper.

A more applied literature has studied the division of surplus between suppliers and retailers in a number of markets, including the television market (e.g., Crawford and Yurukoglu, 2012) and the healthcare market (e.g., Gowrisankaran, Nevo and Town, 2015). This literature relies on a "Nash-in-Nash" approach, in which each pair of firms engages in Nash bargaining, taking as given the agreements reached by all other pairs (see Collard-Wexler, Gowrisankaran and Lee (2017) for a thorough discussion of, and theoretical foundations for, this approach). Although this approach can accommodate downstream competition, it relies on a number of fairly strong assumptions about contracting. In particular, it assumes that, when negotiating their contracts, firms take all other contracts (including other contracts to which they themselves are a party) as given, thus not making use of all the information at their disposal, and that there are gains from trade associated with every potential supply link. These assumptions make it possible to derive, and take to the data, precise and tractable implications regarding the division of surplus over given supply networks. They are, however, less well-suited to studying how firms may affect downstream competition by implementing networks in which some supply links may remain inactive.¹ For example, this approach does not account adequately for the fact that a pair of firms may not find it profitable to rescind a supply link if all other links remain active, but may find it profitable to do so if other supply links are also rescinded. Another limitation of this literature is that it typically constrains payments from retailers to suppliers to be either lumpsum (e.g., Gowrisankaran et al., 2015; Collard-Wexler et al., 2017) or linear without any lumpsum component (e.g., Horn and Wolinsky, 1988; Crawford and Yurukoglu, 2012).

In this paper I advance this literature by combining insights from the literature on network formation with transfers (e.g., Bloch and Jackson, 2007; Jackson, 2008; Bloch and Dutta, 2011) with insights from the literature on vertical contracting with a single supplier (e.g., O'Brien and Shaffer, 1992; McAfee and Schwartz, 1994; Rey and Vergé, 2004) or a single retailer (e.g., O'Brien and Shaffer, 1997; Bernheim and Whinston, 1998). Relative to the "Nash-in-Nash" literature discussed above, this approach provides less precise predictions regarding the division of surplus between upstream and downstream firms, but can shed some light on other important aspects, such as the structure of vertical contracts, the importance of multilateral deviations, the endogenous emergence of narrow supply networks, and the effects of the latter on the intensity of downstream competition.

Overview – The starting point of the analysis is an exploration of how the structure of sup-

¹Liebman (2016) and Ho and Lee (2017) study situations in which health insurance companies commit ex-ante to exclude one or more health care providers from their networks to increase their bargaining leverage. However, they either do not allow for downstream competition (Ho and Lee, 2017) or do not clearly discuss the implications of exclusivity for such competition (Liebman, 2016). Other important differences, discussed in further detail in Section 6.2, are that my framework allows for both upstream and downstream exclusivity and does not allow for ex-ante commitment to exclude.

ply networks affects industry profits.² Networks in which a relatively large number of potential distribution channels remain inactive, because some retailers are excluded or because active retailers distribute different products, have two opposite effects on industry profits. On the one hand, when suppliers and retailers are differentiated, the absence of some product-retailer combinations reduces the demand or willingness to pay expressed by some consumers, thus lowering total industry revenues. On the other hand, it softens downstream competition by reducing the number or increasing the effective differentiation of active retailers. The relative importance of these two effects determines which type of network maximizes industry profits. For example, when retailers are close substitutes, softening downstream competition is more important than preserving retailers' variety. As a result, industry profits are maximized by networks with relatively few retailers or with retailers carrying different products. The opposite is true when retailers are intrinsically highly differentiated.

Although a number of articles has developed similar insights regarding the effects of different supply networks on downstream competition, these articles have either taken such networks as exogenously given (e.g., O'Brien and Shaffer, 1993; Besanko and Perry, 1994) or analyzed very stylized contracting games that, effectively, impose exogenous limits on the types of networks that can arise (e.g., Dobson and Waterson, 1996; Hermalin and Katz, 2013; Nocke and Rey, 2016).³ I advance this literature by developing and solving a full-fledged contracting model that does not preclude any contracting channel, and thus does not exogenously rule out any type of supply network. In particular, I study how externalities arising from upstream and downstream competition may prevent decentralized, bilateral contracting from yielding equilibrium supply networks that maximize industry profits, and how firms can use exclusive contracts to address these externalities, thus affecting equilibrium supply networks, profits and welfare.

The formal model on which I rely is an adaptation of Bloch and Jackson's (2007) model of network formation with transfers to a bilateral oligopoly setting with vertical contracting. In this model, all firms on either side of the market (i.e., all suppliers and all retailers) simultaneously and secretly submit contract proposals to all firms on the other side of the market. A contract proposal includes proposals regarding i) the wholesale price at which the retailer will purchase

²Another factor that may affect industry profits, besides the structure of supply networks, is the extent to which suppliers can soften downstream competition by publicly committing to marginal input prices. As I discuss further below, in this paper I assume that suppliers do not have any ability to commit publicly to such prices and thus to affect equilibrium industry profits through this channel.

³Hermalin and Katz (2013) and Nocke and Rey (2016) consider simple contracting games in which each firm on one side of the market is exogenously matched with one and only one firm on the other side of the market and, therefore, cannot organize multilateral deviations in which it negotiates with, and possibly requires exclusivity from, two or more firms at the same time. Analogously, Dobson and Waterson (1996) assume that each firm is exogenously limited to requiring exclusivity from at most one other firm.

the product from the supplier, ii) exclusivity clauses (if any), and iii) an upfront transfer to be paid by one party to the other at the time the contract is signed. A supplier and a retailer enter a contract only if their proposals regarding all three of these elements are consistent. Once all contracting is concluded, retailers with at least one supply contract compete in prices or quantities in the downstream market.

There are three main differences between my model and that of Bloch and Jackson (2007). First, in Bloch and Jackson's model, and in much of the literature on network formation, the only relevant choice for a pair of players is whether to form or sever a link. In the vertical contracting setting of this paper, instead, each supplier-retailer pair must also specify a wholesale price (see i) above), which affects the price or quantity chosen by the retailer and, through this channel, the payoffs of the network formation game. I show that, in any equilibrium with secret contracts all wholesale prices are equal to marginal cost. Second, the application of the model to a bilateral oligopoly setting implies some inescapable restrictions on payoffs, arising mainly from supplier and retailer substitutability, that make it impossible to rely on some of the assumptions (e.g., nonnegative externalities and link-separability of payoffs) used by Bloch and Jackson to derive some of their results. Finally, as discussed below, in order to deal with the pervasive horizontal externalities in my model, I use Bernheim, Peleg and Whinston's (1987) coalition-proof Nash equilibrium (CPNE) as a solution concept. CPNE allows for multilateral deviations and is a stronger equilibrium concept than the pairwise Nash equilibrium used by Bloch and Jackson (2007), in which firms can only add one new link at a time, or the contract equilibrium proposed by Crémer and Riordan (1987). I solve the model described above by first deriving some general results and then applying them to a bilateral duopoly with linear demand. This allows me to answer the following questions.

First, in an environment without exclusive contracts, can some supply links remain inactive in equilibrium? And under what conditions do equilibrium supply networks maximize industry profits? I show that – with the exception of Bertrand competition between closely substitutable retailers, for which each retailer sells a different product in equilibrium – all potential supply links are generally active in equilibrium, even when supply networks with fewer links would maximize industry profits. This is the case because the inability of firms to use exclusive contracts makes it impossible to prevent certain supplier-retailer pairs from behaving opportunistically and forming new links at the expense of other supplier-retailer pairs.

Second, how does the availability of exclusive contracts affect equilibrium supply networks, welfare and profits? By requiring the consent of broader coalitions of firms for deviations that expand a supply network, exclusive contracts solve the opportunism problem discussed above

and make it possible to support equilibria with narrow networks when these maximize industry profits, although pure-strategy equilibria do not always exist. When exclusive contracts affect the equilibrium structure of supply networks, they always reduce variety and the intensity of downstream competition, resulting in lower consumer and overall welfare. Moreover, the availability of exclusive contracts, by affecting the disagreement payoffs of suppliers and retailers differently, redistributes profits from retailers to suppliers, even when such contracts are not adopted in equilibrium.

Finally, how do constraints on the firms' ability to use upfront transfers or long-term contracts and to create instantaneously new supply links at the network formation stage affect equilibrium outcomes? In the baseline model discussed above there are no constraints on such ability. As a result, the division of the profits generated by new supply links takes place at the same time that the network is formed and is not affected by hold-up problems. Instead, if firms must form a complete supply network, for example by making specific investments, *before* starting to negotiate supply contracts and agreeing on any transfers, the division of profits resulting from ex-post bargaining is affected by hold up, as in Lee and Fong (2013) and Rey and Vergé (2016) (discussed at the end of this introduction). I show that, relative to an environment with upfront transfers, ex-post bargaining with hold up makes it easier to support equilibria with inactive supply links because firms find it more difficult to share the joint profits from deviations that form new supply links. This has ambiguous effects on industry profits, but unambiguously reduces consumer and overall welfare.

From a methodological point of view, solving the model of simultaneous contracting with transfers presented in this paper poses a number of challenges. One such challenge lies in applying the CPNE solution concept to an environment with transfers. CPNE requires coalitional deviations to be immune to further deviations by subcoalitions. When firms can use transfers, this implies that the transfers that support successive deviations are related, and one must keep track of their relationships. Specifically, the transfers that make a given deviation profitable for all firms involved in that deviation may depend on the transfers that made the previous deviation in the sequence profitable, and so on, potentially all the way back to the transfers that support the candidate equilibrium. I tackle this issue by proposing a general algorithm that I subsequently use to characterize the CPNE of a tractable bilateral duopoly model.

Another challenge is the determination of equilibrium wholesale prices. As is well-known, in the presence of downstream competition equilibrium wholesale prices depend on the extent to which a supplier can publicly commit to his offers and, if public commitment is not available, on the beliefs held by retailers about that supplier's dealings with rival retailers. In this paper

I assume that firms cannot publicly commit to their contract proposals and show that all CPNE wholesale prices are equal to marginal cost. Similar results have already been established by the existing literature in different settings (e.g., a single supplier or a single retailer) and/or for different equilibrium concepts (e.g., PBE or contract equilibrium). Here I extend those findings to the CPNE of a bilateral oligopoly. I also show that, by requiring multilateral deviations to be self-enforcing, CPNE avoids the issues identified by Rey and Vergé (2004) for the existence of perfect Bayesian equilibrium wholesale prices with Bertrand downstream competition.

Other related literature – This paper is also related to Rey and Vergé (2016) and Lee and Fong (2013). As in this paper, Rey and Vergé (2016) allow firms to engage in multiple bilateral contract negotiations. However, contrary to this paper, they do not allow for the use of transfers or exclusive contracts at the stage in which supply relations are initially formed, limiting their attention to the case in which the surplus from any relation can only be divided ex-post under conditions of hold up. Lee and Fong (2013) present a model of dynamic network formation in which, at any point in time, firms can only bargain over the existing network and, like in Rey and Vergé (2016), cannot create new supply links. However, they allow firms to create new supply links in subsequent periods. When the cost from doing so is low and the time between periods is short, Lee and Fong's (2013) environment approaches the environment without hold up that I study in this paper. Notwithstanding this similarity, their model and mine are quite different. Lee and Fong (2013) emphasize intrinsically dynamic aspects, such as the response of networks to shocks in the presence of adjustment costs, and simplify other aspects by, e.g., assuming that firms can only use lump-sum transfers and that exclusive contracts are not available. Instead, I adopt a static model of simultaneous contracting and emphasize the role played by the structure of vertical contracts and the degree of supplier and retailer differentiation. By relying on coalition-proof Nash equilibrium, I also propose a more systematic refinement of the set of equilibria than Lee and Fong (2013). Finally, I provide an in-depth analysis of the role played by exclusive contracts, which are not addressed in Rey and Vergé (2016) and Lee and Fong (2013), and are more naturally studied in an environment like mine, in which firms can use upfront payments to purchase exclusivity.

Organization of the paper – Section 2 introduces a formal model with upfront transfers. Section 3 presents the solution method and some general results. Sections 4 through 6 study supply networks with and without exclusive contracts in a bilateral duopoly model with linear demand. Section 7 explores the effects of ex-post bargaining and hold up. Section 8 concludes. All proofs are in Appendix A. Supplemental material is contained in Appendix B (available online).⁴

⁴This online appendix will be available shortly at https://sites.google.com/site/paoloramezzana/.

2 Model and equilibrium concept

This section first introduces a model of contracting and competition in bilateral oligopoly and then discusses why Bernheim, Peleg and Whinston's (1987) coalition-proof Nash equilibrium (CPNE) is an appropriate solution concept for this model.

2.1 Model

There are $S \ge 2$ suppliers, each producing a different product at constant marginal cost c. The products are imperfect substitutes and are distributed to consumers by $R \ge 2$ differentiated and competing retailers at no additional costs besides their payments to suppliers (introduced further below). With a slight abuse of notation, S and R denote both the number and the set of suppliers and retailers, respectively. Let q_{sr} denote the quantity and p_{sr} the retail price of product $s \in S$ sold by retailer $r \in R$, with $q \in \mathbb{R}^{S \times R}$ and $p \in \mathbb{R}^{S \times R}$ denoting the vectors of all quantities and prices. The direct demand for product s at retailer r is $q_{sr} = D_{sr}(p)$, with inverse demand $p_{sr} = P_{sr}(q)$. In Section 4, I will introduce a linear demand system that allows me to parametrize the degree of supplier and retailer differentiation.

Given this environment, I study a two-stage game, in which first suppliers and retailers negotiate bilateral supply contracts and then retailers compete in the downstream market. Specifically, in the first stage suppliers and retailers engage in the following simultaneous contracting game with transfers, which is an adaptation of the game of network formation with transfers introduced by Bloch and Jackson (2007). Denote by $K = \{S, R\}$ one side of the market (i.e. the set of suppliers or the set of retailers) and by $K' = \{S, R\}$, $K' \neq K$, the other side of the market. Each firm $i \in K$ announces a contract proposal $x_i^j = \langle t_i^j, w_i^j, \theta_i^j \rangle$ for each firm $j \in K'$, where $t_i^j \geq 0$ and $w_i^j \geq 0$ are, respectively, the proposed transfer and wholesale price that the retailer must pay to the supplier, and θ_i^j the type of exclusive arrangement, if any, that governs the relationship between *i* and *j*. Exclusivity arrangements may involve one-way exclusivity, with which *i* commits to be exclusive to *j* or vice versa, or mutual exclusivity and will be discussed in further detail in Section 6. I assume that firms cannot publicly commit to their contract proposals. I discuss the details of this assumption below, after having discussed the equilibrium concept adopted.

Denoting by *s* the supplier and by *r* the retailer in the pair (i, j), *s* and *r* reach an agreement if $w_s^r = w_r^s$, $\theta_s^r = \theta_r^s$, and $t_r^s \ge t_s^r$, where the latter condition means that the retailer is willing to offer a transfer t_r^s that meets the transfer request t_s^r of the supplier. When *s* and *r* reach an agreement, a contract with $w_{sr} = w_s^r = w_r^s$, $\theta_{sr} = \theta_s^r = \theta_r^s$ and $t_{sr} = t_r^s$ enters into effect and a supply link, denoted by $\ell_{sr} = 1$, is formed.⁵ Otherwise no link is formed and $\ell_{sr} = 0$. The collection

⁵Note that in any equilibrium it will always be $t_r^s = t_{s'}^r$ otherwise either *s* or *r* would have a profitable

 $\ell = \langle \ell_{sr} \rangle_{s \in S, r \in R}$ of all supply links, or lack thereof, gives rise to a supply network $g = g(\ell)$.

When bilateral contracting is concluded, retailers observe the resulting supply network g, but not the wholesale prices in the contracts signed by other retailers, and compete in the downstream market in stage 2. Since the general principles of the analysis that follows apply equally well to Cournot or Bertrand downstream competition, I allow for either mode of competition. I assume that for any supply network g and profile of wholesale prices $w \in \mathbb{R}^{S \times R}$, downstream competition results in a unique equilibrium profile of retail prices p(g, w) and quantities q(g, w). Given a profile of contracts $x = \langle t, w, \theta \rangle$, and the resulting supply network g, the payoffs of supplier s and retailer r are therefore, respectively,

$$\pi_{s}(g,t,w) = \sum_{r \in R} \ell_{sr} \left[t_{sr} + (w_{sr} - c) q_{sr}(g,w) \right],$$
(1)

$$\pi_r(g,t,w) = \sum_{s \in S} \ell_{sr} \left[(p_{sr} - w_{sr}) q_{sr}(g,w) - t_{sr} \right].$$
(2)

Throughout the paper it will also be helpful to keep track of the total vertical profits (gross of any payment to suppliers) generated by retailer *r*, which are given by

$$\Pi_{r}(g,w) = \sum_{s \in S} \ell_{sr}(g) \left[p_{sr}(g,w) - c \right] q_{sr}(g,w) \,. \tag{3}$$

For future reference it is important to note that, because of downstream competition, total industry profits, $\sum_{r \in R} \prod_r (g, w)$, are not necessarily maximized when all suppliers trade with all retailers. Networks in which some supply links are not active, such as downstream monopoly or pairwise exclusivity, sacrifice some variety but reduce the intensity of downstream competition between retailers and may thus yield higher overall industry profits.

In order to derive some of the results in subsequent sections, it is also helpful to ensure that secret changes in w_{sr} , i.e., changes in w_{sr} that are observed only by retailer r, cause well-behaved responses in the derived demand for product s. This is guaranteed by the following assumption.

Assumption 1 (Derived demand) For all g, and all $s \in S$ and $r \in R$ for which $\ell_{sr}^g = 1$, secret changes in w_{sr} have the following effects on the derived demand for product s:

$$\frac{dq_{sr}}{dw_{sr}} < 0, \quad \frac{dq_{sj}}{dw_{sr}} \ge 0 \text{ for all } j \in \mathbb{R} \text{ with } \ell_{sj}^g = 1 \text{ and } j \neq r, \text{ and } \frac{dq_s}{dw_{sr}} = \sum_{j \in \mathbb{R}} \ell_{sr}^g \frac{dq_{sj}}{dw_{sr}} < 0$$

A change in w_{sr} that is unobserved by other retailers $j \neq r$ can affect directly only the vector of quantities q_r (when competition is Cournot) or retail prices p_r (when competition is Bertrand) chosen by retailer r for the products that she carries. When competition is Cournot, no other

unilateral deviation. It is, therefore, immaterial whether the transfer paid in equilibrium is that offered by the retailer (i.e., $t_{sr} = t_r^s$, as assumed here) or that requested by the supplier (i.e., $t_{sr} = t_s^r$).

changes in quantities occur and thus $dq_{sj}/dw_{sr} = 0$ for all $j \neq r$. The only "bite" of Assumption 1 in this case is therefore to ensure that $dq_{sr}/dw_{sr} < 0$ and thus $dq_s/dw_{sr} < 0$. When competition is Bertrand, the changes in p_r induced by changes in w_{sr} cause instead the quantities sold by other retailers to change, even if the retail prices charged by those other retailers do not change, so that $dq_{sj}/dw_{sr} > 0$ for all $j \neq r$. In this case Assumption 1 ensures that the direct effect, $dq_{sr}/dw_{sr} < 0$, dominates the indirect effects, $\sum_{j\neq r} dq_{sj}/dw_{sr} > 0$, of a change in w_{sr} on the demand for product *s* and thus that the overall market demand for product *s* is downward sloping in w_{sr} . Assumption 1 holds for the linear demand system that will be introduced in Section 4.

2.2 Equilibrium concept

When suppliers and retailers act unilaterally and cannot coordinate their strategies (i.e., their contract proposals), the contracting game described above has a large number of Nash equilibria, many of which are supported by coordination failures. For example, as in all games of network formation, vertical coordination failures may give rise to Nash equilibria in which a supplier *s* and a retailer *r* do not form a supply link, even though it would be jointly profitable for them to do so. Moreover, in the simultaneous contracting model studied in this paper, even when *s* and *r* manage to form a link, they may still fail to coordinate on the terms of the contract, w_{sr} or θ_{sr} , that would maximize their joint profits. These vertical coordination failures may be avoided by adopting "bilateral" solution concepts, such as contract equilibrium (e.g., Crémer and Riordan, 1987; O'Brien and Shaffer, 1992), in which each *sr* pair maximizes its joint profits, given all other contracts (including other contracts to which *s* and *r* are a party). Although such equilibrium concepts are appropriate in some settings, they cannot address two aspects that play an important role in this paper.

First, contract equilibrium is not sufficient to rule out the horizontal coordination failures that may arise in a setting with bilateral oligopoly and exclusive contracts. For example, consider the bilateral duopoly model illustrated in Figure 1, in which solid lines represent active supply links and dashed lines represent potential supply links that are inactive.⁶ Focus on a candidate equilibrium in which *S*1 and *R*1 and *S*2 and *R*2 adopt mutually exclusive contracts, resulting in the pairwise exclusive supply network shown in Figure 1(c). Assume that the double common agency supply network in Figure 1(a) and/or the mixed supply network in Figure 1(b) yield higher overall industry profits than a pairwise exclusive network. A deviation from pairwise exclusivity to one of these two networks could benefit all firms but cannot be achieved unilaterally

⁶This bilateral duopoly model will also form the basis for other examples introduced below and for the analysis of Sections 4 through 7.



Figure 1: Supply networks in a bilateral duopoly.

or bilaterally, as it requires coordination among all four firms. This is the case because, given the mutually exclusive contracts that support the candidate pairwise exclusive equilibrium, *S*2 and *R*1 cannot, individually or bilaterally, form a supply link while maintaining their other supply links. For this to be possible, *S*1 must forgo the exclusivity he demands of *R*1, and *R*2 must forgo the exclusivity she demands of *S*2. Moreover, a deviation to double common agency also requires the pairs S1 - R2 and S2 - R1 to coordinate the formation of the two new diagonal links. One can also construct examples, with our without exclusive contracts, where firms may profit from coordinating deviations in which they rescind more than one existing link at a time, instead of creating more than one new link at a time.

Second, as already discussed in the introduction, contract equilibrium does not allow for multilateral deviations in which firms modify more than one contract at a time. This is particularly problematic in a setting with exclusive contracts as the one of this paper, because it makes sense for a supplier s and a retailer r to enter a (one-way or two-way) exclusive contract only if at least one of them can modify (i.e., rescind) its other contracts. Besides this almost mechanical point, many deviations in which one or more firms attempt to modify a supply network are profitable only if such deviations entail modifying more than one contract at a time. For example, starting from a candidate equilibrium with double common agency as in Figure 1(a), R1 may find it profitable to sign up *S*1 to exclusivity if he can also sign up *S*2 to exclusivity, thus implementing a downstream monopoly as in Figure 1(d), but may not find it profitable to do so if he must continue in its nonexclusive contract with *S*2 (as mandated by contract equilibrium), thus implementing a mixed network as in Figure 1(b).

To avoid the issues discussed above, I adopt as a solution concept Bernheim, Peleg and Whinston's (1987) coalition-proof Nash equilibrium (CPNE), which allows for multilateral deviations and (nonbinding) coordination. A profile of strategies (i.e., of contract proposals in this paper) constitutes a CPNE if it is immune to mutually profitable and self-enforcing *nonbinding* agreements to deviate by any coalition of firms, taking as given the strategies of firms that do not participate in the coalition. An agreement is self-enforcing if it is itself immune to further selfenforcing deviations by subcoalitions, where these further deviations are also required to be self-enforcing, and so on (see Bernheim, Peleg and Whinston (1987) for a formal definition).⁷ CPNE imposes some degree of consistency on deviations, ruling out deviations that should not be considered "credible" (i.e., self-enforcing).

Finally, even though firms cannot publicly commit to their contract proposals, I do not need to specify the beliefs held by firms involved in deviations. This is because in a CPNE of the game of simultaneous contracting introduced in Section 2.1 no firm participating in a deviation can unilaterally modify its contracts with firms that do not participate in that deviation. As discussed further in Section 3.1, this feature of CPNE has implications that are analogous to those of contract equilibrium for the determination of equilibrium wholesale prices.

3 Some general properties of equilibria

One can characterize the CPNE of this model in two stages, by first characterizing the selfenforcing profile of wholesale prices w(g) for any network g, and then studying which networks g can be supported as equilibria.

3.1 Equilibrium wholesale prices

When Assumption 1 on derived demand holds, one can establish the following result, which takes as given the supply links in a given supply network *g* and characterizes wholesale prices in the contracts governing those links. This (intermediate) result is, therefore, obtained not by asking whether there exist deviations to contracts that would implement different supply networks,

⁷Although the definition is recursive, it avoids cycles because it limits further deviations (but not the original deviation from the candidate equilibrium) to subcoalitions and has thus a natural end point. This aspect will be discussed in further detail in Section 3.2.

but rather whether there exist deviations to contracts that implement the same supply network with different wholesale prices.

Proposition 1 (Equilibrium wholesale prices) When firms cannot publicly commit to their contract proposals, for any supply network g there always exists a unique coalition-proof Nash equilibrium profile of wholesale prices w(g), with w(g) = c, regardless of the mode of downstream competition.

Proposition 1 extends results obtained in different settings and for different equilibrium concepts by the vertical contracting literature (briefly discussed below) to the CPNE of the bilateral oligopoly model studied in this paper. Its logic can be understood in two steps. First, since any jointly profitable bilateral deviation by *s* and *r* to a different w_{sr} is self-enforcing, a CPNE must be immune to any such deviation and is therefore also a contract equilibrium. By extending O'Brien and Shaffer's (1992) analysis of contract equilibria with a single supplier to a setting with multiple suppliers, one can therefore prove that in any CPNE it must be $w_{sr} = c$ for all *s* and *r*. If this were not the case, any supplier-retailer pair for which $w_{sr} > c$ could profitably engage in a bilateral deviation to $w_{sr} = c$, which would allow this pair to appropriate some of the retail margins of other retailers. Second, the above implies that any multilateral deviation that changed the wholesale prices of two or more retailers away from w = c at the same time would never be self enforcing, since it would always be blocked by a further self-enforcing bilateral deviation to marginal cost pricing, regardless of the mode of downstream competition.

This last aspect is what distinguishes Proposition 1 from existing literature, especially for the case of Bertrand downstream competition. As shown by Rey and Vergé (2004), when retailers compete à la Bertrand, are sufficiently close substitutes and hold passive beliefs, multilateral deviations in which suppliers raise the wholesale prices of two or more retailers above marginal cost at the same time may become profitable. If one adopts an equilibrium concept such as perfect Bayesian equilibrium, which allows for such multilateral deviations without requiring them to be self enforcing, this implies that pure-strategy equilibria may fail to exist, because there exist profitable deviations both when all wholesale prices are equal to marginal cost and when some of them are not. The vertical contracting literature typically addresses these equilibrium existence issues by limiting attention to contract equilibrium concepts that allow only for bilateral deviations (see, e.g., O'Brien and Shaffer (1992) and McAfee and Schwartz (1994) for models without upstream competition and Rey and Vergé (2016) for a bilateral oligopoly model with upstream competition).⁸ The CPNE solution concept adopted in this paper

⁸The existence issue does not arise with Cournot downstream competition because in that case multilateral deviations do not present any advantage relative to bilateral deviations.

does not face this stark choice, as it overcomes the existence problems associated with Bertrand competition while allowing for multilateral deviations.

3.2 Equilibrium supply networks

Having shown that, when firms cannot commit to their wholesale price proposals, equilibrium wholesale prices are equal to marginal cost in all possible networks, the next step is to determine which of these networks can be supported as equilibria by some profile of transfers. As will become clear below, when a network g can in fact be supported as an equilibrium there typically exists a (possibly broad) range of transfers t^g for which this is the case. The model is, therefore, better suited to shedding light on what types of supply networks arise as equilibria under different conditions, rather than to predicting exactly how profits will be split between firms in those equilibria. As the division of profits resulting from t^g is irrelevant for consumer and overall welfare, because it does not directly affect product or retailer variety or the intensity of downstream competition, I do not view this as a significant shortcoming of the model.⁹ Moreover, as shown in Section 6.2, certain changes in the environment, such as the availability of exclusive contracts, cause the range of suppliers or retailers' equilibrium profits to shift entirely to the right or the left of their initial range. When this is the case, the approach taken in this model is sufficient to determine unambiguously the distributional effects of those changes.

In order to predict the exact level of t^g one would have to study a different model in which suppliers and retailers engage in some form of coalitional bargaining at the contracting stage.¹⁰ However, most existing models of coalitional bargaining – see, e.g., Shapley (1953) for a cooperative model and Chatterjee et al. (1993) for a noncooperative model – assume that the grand coalition always forms and that there are no externalities between coalitions; or, when they allow for externalities between coalitions (e.g., Ray and Vohra, 1999), they rely to a large extent on symmetry among all players to obtain tractable results.¹¹ These assumptions do not describe well the environment studied in this paper, in which externalities resulting from downstream competition and exclusive contracts, as well as asymmetries between suppliers and retailers,

⁹It should, however, be noted that the division of profits among firms can have welfare effects if it affects the investment incentives of firms, an aspect that is not modeled in this paper.

¹⁰Such a model would be different, and have different implications, from the model studied in Section 7, in which firms first form a supply network in the absence of upfront transfers and then engage in ex-post bilateral bargaining over the existing supply network. Whereas hold-up plays no role in the coalitional bargaining game with upfront transfers described in the text above, it plays a crucial role in the model of ex-post bilateral bargaining of Section 7.

¹¹For a discussion of these issues see Bloch and Dutta (2011) and Maskin (2003). In particular, Maskin argues that the assumption that the grand coalition always forms is one of the main shortcomings of noncooperative game theory.

play a crucial role. In light of this, developing a full-fledged coalitional bargaining game that is well-suited to this complex environment, only for the purpose of obtaining precise predictions regarding transfers, would be well beyond the scope of this paper and is not attempted here.¹²

Having clarified the scope of the analysis, I can proceed with a characterization of equilibrium supply networks. I first introduce formal definitions of self-enforcing agreements and of CPNE supply networks in the specific model with transfers of this paper, and then derive some helpful results that allow me to make these definitions operational.

Denote by $Z_{g \to h}$ the set of all coalitions that, starting from a network g, can implement a network $h \neq g$ without requiring the consent of firms outside the coalition (i.e., without need-ing firms outside the coalition to modify their strategies). One can then define a self-enforcing nonbinding agreement as follows.

Definition 1 (Self-enforcing agreement) A nonbinding agreement among the members of coalition $Z \subseteq S \cup R$ to implement a supply network g with transfers t^g is self-enforcing if there does not exist any other self-enforcing agreement among the members of any subcoalition $Z' \subseteq Z$, $Z' \in \mathcal{Z}_{g \to h}$, that implements a supply network $h \neq g$ with transfers t^h such that, for all $i \in Z'$,

$$\pi_i(h, t^h) > \pi_i(g, t^g) \tag{4}$$

A few aspects of Definition 1 are worth noting. First, the definition is recursive: an agreement is self-enforcing if and only if it cannot be improved upon by another self-enforcing agreement. The problem remains, however, well defined because deviations from agreements are restricted to subcoalitions Z' of the coalition Z that reaches the original agreement. This limits the number of successive deviations that one needs to consider and a solution can be reached in a finite number of steps. In the bilateral duopoly applications discussed in Sections 5 through 7 this number of steps is generally small and the problem remains tractable.

Second, the ability of firms to implement a new supply network *h* starting from a supply network *g*, and the size and composition of the coalitions that can do so, captured by $Z_{g \to h}$, depend on a number of factors, such as the extent to which firms are allowed to communicate with one another and whether exclusive contracts are allowed or not. For example, it is generally

¹²One can, however, conjecture that an efficient coalitional bargaining game in which coalitions always reach agreements that are mutually profitable relative to alternative *self-enforcing* outcomes would yield the same equilibrium networks as the simultaneous contracting game that I study in this paper. Specifically, this should be the case for bargaining games in which firms i) can continue to make proposals to each other until all possibilities have been explored and ii) cannot credibly commit ex-ante to deal with a limited number of counterparts in order to elicit better offers from those counterparts. If this conjecture is correct, the analysis in this paper characterizes the bargaining set to which the outcomes of such efficient bargaining games must belong.

more difficult for firms to implement a new supply network when nonbinding communication is partially restricted (e.g., when communication between firms on the same side of the market is prohibited). As for exclusive contracts, their adoption makes it mode difficult to deviate from a given network g to broader networks h that add new supply links to g without rescinding existing links. Such deviations would require consent from broader coalitions than in the absence of exclusive contracts, because some firms that have exclusive rights in network g must consent to the expansions of the network from g to h. These aspects will be discussed in further detail in Sections 5 and 6, where they will play important roles.

Building on Definition 1, one can define a CPNE supply network as follows.

Definition 2 (CPNE with transfers) A supply network g^* with transfers t^{g^*} constitutes a CPNE if and only if there does not exist any self-enforcing nonbinding agreement among the members of any coalition $Z \in \mathcal{Z}_{g^* \to g}$ that implements a supply network $g \neq g^*$ with transfers t^g such that $\pi_i(g, t^g) > \pi_i(g^*, t^{g^*})$ for all $i \in Z$.

To avoid confusion, note that the initial deviations from a candidate equilibrium supply network g^* to a different network g in Definition 2 can be carried out by any coalition $Z \in \mathcal{Z}_{g^* \to g}$, whereas successive deviations from g can only be carried out by subcoalitions $Z' \subseteq Z$ (see Definition 1).

Definitions 1 and 2 can be made operational for use in the applications presented in Sections 5 through 7 as follows. Since, by Proposition 1, w(g) = c for all networks g, (1) and (2) take the following simple forms.

$$\pi_s(g, t^g) = \sum_{r \in R} \ell^g_{sr} t^g_{sr}$$
(5)

and

$$\pi_r(g, t^g) = \Pi_r^g - \sum_{s \in S} \ell_{sr}^g t_{sr}^g, \tag{6}$$

where $\Pi_r^g \equiv \Pi_r(g,c)$ is given by (3) and denotes the gross vertical profits generated by retailer r in supply network g when all wholesale prices are equal to marginal cost. Using these expressions, one can establish the following result regarding the mutual profitability of coalitional deviations from any network g to any network h.

Lemma 1 Given a network g with transfers t^g , there exists a network $h \neq g$ and profile of transfers t^h such that $\pi_i(h, t^h) > \pi_i(g, t^g)$ for all $i \in Z$ if and only if

$$\sum_{r\in\mathbb{Z}} \left[\Pi_r^h - \Pi_r^g \right] > \sum_{s\in\mathbb{Z}} \sum_{r\notin\mathbb{Z}} \left(\ell_{sr}^g - \ell_{sr}^h \right) t_{sr}^g - \sum_{r\in\mathbb{Z}} \sum_{s\notin\mathbb{Z}} \left(\ell_{sr}^g - \ell_{sr}^h \right) t_{sr}^g \tag{7}$$

Lemma 1 is used below to construct an algorithm to solve for the CPNE of the model. Before doing so, however, it may be helpful to discuss a few intuitive aspects that play a role in that solution. The left-hand side of (7) represents the incremental gross vertical profits generated by the retailers that participate in a deviation from network *g* to network *h*. These incremental profits can be positive or negative. The right-hand side represents, instead, the net loss of transfers from firms with which the deviating coalition rescinds existing links or forms new links. For example, with reference to Figure 1, if *S*1 and *R*1 form a coalition and deviate from Figure 1(a) to Figure 1(c) by rescinding their links with *R*2 and *S*2, respectively, the right-hand side of (7) corresponds to the difference between the transfer that *S*1 receives from *R*2 in Figure 1(a) and the transfer that *R*1 pays to *S*2 in Figure 1(a).¹³ In general, this net loss can also be positive or negative. Lemma 1 states that there exists a mutually profitable deviation from a network *g* with transfers *t*^g to a different network *h* if and only if the incremental vertical profits exceed the net loss in transfers.

Of particular interest is the question of whether a coalition Z can engage in mutually profitable deviations from a network g that maximizes industry profits. In this respect, two types of deviations play an important role in the applications that follow. In the first type of deviations the members of the deviating coalition Z do not modify their links (if any) with firms outside *Z*, i.e., $\ell_{sr}^g = \ell_{sr}^h$ if $s \notin Z$ or $r \notin Z$ and the right-hand side of (7) is thus equal to zero for any t^{g} . If the number of retailers involved in the deviation is strictly less than the total number of retailers *R* and $\sum_{r \in Z} \prod_{r=Z}^{h} \sum_{r \in Z} \prod_{r=Z}^{g} \prod_$ even though $\sum_{r \in R} \prod_{r=R}^{h} \prod_{r=R}^{g} \prod_{r=R}^{g}$, because of the negative externalities imposed on the firms that do not participate in the deviation. As an example, which foreshadows the analysis in Section 5, refer to the pairwise exclusive network in Figure 1(c) and assume that exclusive contracts are not allowed. Consider a deviation in which the two firms in coalition $Z = \{R1, S2\}$ form a link with each other while preserving their existing links with other firms, thus causing the market to move to the mixed network in Figure 1(b). This deviation does not alter any links with firms outside coalition Z and, by condition (7), is profitable if and only if it increases the vertical profits generated by the only retailer in the coalition, R1, even if total industry profits, i.e., the sum of the vertical profits generated by R1 and R2, are lower in Figure 1(b) than in Figure 1(c). As will be discussed in Section 6, exclusive contracts can render such a deviation mutually unprofitable, by making the consent of all four firms necessary (i.e., by restricting the set $Z_{g \to h}$ of coalitions

¹³In this context a coalition is simply a set of firms discussing whether and how to coordinate their strategies (i.e., their transfer and exclusivity proposals t and θ) to achieve a certain outcome, taking as given the strategies of firms outside the set. Firms that belong to a coalition can also continue, stop, or start dealing with firms outside the coalition by deciding to agree or not to the *given* transfer and exclusivity proposals of those firms. In other words, a coalition formed for the purpose of coordinating the strategies of its members does not necessarily coincide with the set of firms with which its members have a supply link.

that can implement this deviation).

The second type of deviations of interest is one in which the members of Z rescind their links with some firms that do not participate in the deviation, i.e., in which $\ell_{sr}^g = 1$ and $\ell_{sr}^h = 0$ for some $s \in Z$ and $r \notin Z$ or some $s \notin Z$ and $r \in Z$. This type of deviations is mutually profitable for the members of Z if, given an initial profile of transfers t^g , the firms dropped from network g appropriate more than their contribution to g relative to some other network h that can be implemented without their participation. Importantly, the profile of transfers t^g may not provide enough degrees of freedom to ensure that *all* firms extract less than their contribution to a given network g relative to *every* alternative network, and thus g may always be vulnerable to at least one mutually profitable (though not necessarily self-enforcing) deviation.¹⁴ This is, for example, the case for some parameter values in the bilateral duopoly model with exclusive contracts studied in Section 6, for which pure-strategy equilibria may fail to exist.

As already mentioned above, Lemma 1 can be used to construct the following algorithm to solve for the CPNE of the contracting model with transfers introduced in Section 2. For any possible candidate equilibrium supply network *g*:

- Step 1 Determine the set of all alternative networks *h* that can be implemented by any coalition $Z \in \mathcal{Z}_{g \to h}$. If there exists a profile of transfers t^g such that (7) fails for all *h* and $Z \in \mathcal{Z}_{g \to h}$ (i.e., such that no deviation is mutually profitable), one can stop and conclude that *g* constitutes a CPNE (and, in fact, a strong equilibrium). If instead no such t^g exists, for each t^g and each mutually profitable deviation by any coalition $Z \in \mathcal{Z}_{g \to h}$ implementing a network $h \neq g$, determine the range of transfers $I^h(t^g) = \left\{t^h : \pi_i(h, t^h) > \pi_i(g, t^g), \forall i \in Z\right\}$ that can make each member of the deviating coalition better off and move to Step 2.
- **Step 2** For each network *h* identified above, determine all the alternative networks *k* that can be implemented by any *subcoalition* Z' of Z and verify whether, for each t^g , there exists any $\tilde{t}^h \in I^h(t^g)$ such that (7) fails for all *k* and $Z' \in \mathcal{Z}_{h \to k}$, $Z' \subseteq Z$, when one substitutes Z' for Z, *h* for *g*, *k* for *h*, and \tilde{t}^h for t^g (i.e., such that there exists no mutually profitable deviation from *h* to *k*). If there exists such a \tilde{t}^h one can stop here and conclude that the original deviation to *h* is self-enforcing for each t^g and thus that *g* is not a CPNE. If instead there does not exist such a \tilde{t}^h for at least some t^g , then some deviations from *h* to *k* are mutually profitable and one should apply Step 2 to *k* to determine whether any of these deviations are self-enforcing. If the answer is yes, the original deviation to *h* is not self-enforcing and

¹⁴See also Bloch and Jackson (2007) for a discussion of this issue.

g is a CPNE. If the answer is no, the original deviation to *h* is self-enforcing and *g* is not a CPNE.

The solution algorithm introduced above is well defined for any arbitrary numbers of suppliers, *S*, and retailers, *R*. In particular, since successive deviations are limited to subcoalitions, it always converges to an end point. However, using it to solve a model with more than a few firms on each side of the market would be unwieldy because the numbers of possible supply networks and coalitions grow exponentially with *S* and *R*. Related to this, large values of *S* and *R* would also give rise to an unmanageably large number of deviations from deviations that would need to be checked, i.e. of iterations of Step 2 above. For this reason, in the rest or the paper I restrict attention to a more tractable bilateral duopoly model with two symmetrically differentiated suppliers and two symmetrically differentiated retailers that satisfies the assumptions laid out in Section 2.

4 A bilateral duopoly model

The possible types of supply networks that can arise in bilateral duopoly are listed below, together with the maximum vertical profits Π_r^g that each retailer *r* can generate in supply network *g* when all retailers obtain products at wholesale prices w(g) = c and compete in the downstream market.

- *Double common agency* (denoted by g = dca and illustrated in Figure 1(a)). Both retailers deal with both suppliers. Given the symmetry of the model, each retailer generates the same vertical profits $\Pi_r^{dca} = \Pi^{dca}$.
- *Mixed network* (g = mix, Figure 1(b)). One of the retailers deals with both suppliers, while the other retailer only deals with one supplier. The vertical profits generated by the retailer that deals with both suppliers are Π^{mix2} , whereas those of the retailer that only deals with one supplier are Π^{mix1} .
- *Pairwise exclusivity* (g = pe, Figure 1(c)). Each retailer deals with a different supplier and generates the same vertical profits Π^{pe}_r = Π^{pe}.
- *Downstream monopoly* (g = dm, Figure 1(d)). Both suppliers only deal with the same retailer, thus excluding the other retailer. The active retailer generates vertical profits Π^{dm} , whereas the excluded retailer generates zero vertical profits.

- Upstream monopoly (g = um, Figure 1(e)). Both retailers only deal with the same supplier, excluding the other supplier. Each retailer generates the same vertical profits $\Pi_r^{um} = \Pi^{um}$.
- Bilateral monopoly (g = bm, Figure 1(f)): A retailer and a supplier only deal with each other, while the other retailer and supplier are excluded. The active retailer generates vertical profits Π^{bm}, whereas the excluded retailer generates zero vertical profits.

Although one could in principle use a fairly general demand system to rank unambiguously the vertical profits generated by retailers under some of the supply networks listed above, this is not the case for all of these networks.¹⁵ Moreover, even if one could obtain an ordinal ranking of vertical profits for all supply networks, this would be insufficient because the analysis in subsequent sections involves linear functions of these vertical profits and requires these profits to be cardinally comparable. In the interest of concreteness, therefore, I use the following (inverse) linear demand system, which allows me to parametrize the degrees of supplier and retailer differentiation

$$p_{sr} = v - (q_{sr} + aq_{s'r}) - b(q_{sr'} + aq_{s'r'}),$$
(8)

where v > c and $a, b \in [0, 1]$.¹⁶ Lower values of a and b indicate, respectively, higher supplier and retailer differentiation, with a = 0 and b = 0 corresponding to the extreme case in which, respectively, suppliers or retailers are completely independent in demand and a = 1 and b = 1to the extreme case in which they are perfect substitutes. Note that supplier differentiation and retailer differentiation are "cumulative", in that the coefficient ab on $q_{s'r'}$ in (8) is smaller than the coefficients a and b on $q_{s'r}$ and $q_{sr'}$, respectively. A complete solution of the downstream market game played by retailers in stage 3 with this demand system and w(g) = c under both Cournot and Bertrand competition is presented in an online appendix (enclosed with this submission). In the body of the paper I will rely directly on the values of the vertical profits Π^g resulting from that solution.

¹⁵For example, under mild assumptions on demand, one can generally conclude that the profits generated by a retailer are decreasing in the number of products carried by the other retailer, which implies, e.g., $\Pi^{mix2} \ge \Pi^{dca}$ and $\Pi^{dm} \ge \Pi^g$ for all $g \ne dm$; or that, given the number of products carried by the two retailers, profits are higher when the retailers carry different products, which implies $\Pi^{pe} \ge \Pi^{um}$. However, other comparisons are more ambiguous. For example, one cannot necessarily conclude that the equilibrium vertical profits generated by a retailer increase in the number of products carried by that retailer, because the equilibrium prices charged by the other retailer are likely to fall, or the quantities likely to increase, when the first retailer introduces new products. This implies that, in general, there is no unambiguous comparison between, e.g., Π^{mix2} and Π^{pe} or Π^{dca} and Π^{mix1} .

¹⁶See the online appendix for a derivation of this demand system from underlying consumer preferences and for a verification that it satisfies Assumption 1.

This framework is used in Figure 2 to characterize the supply networks that maximize total industry profits, conditional on retailers competing in the downstream market. Comparisons between these networks and the networks that arise in equilibrium will prove particularly helpful in the sections that follow, as they will provide insights into the nature and magnitude of the externalities that prevent equilibrium networks from maximizing industry profits. Figure 2 divides



Figure 2: Supply networks that maximize industry profits.

the unit square representing all possible combinations of supplier substitutability, *a*, and retailer substitutability, *b*, into three regions and shows that industry profits are maximized by a different supply network in each of these regions. Specifically, denoting with $\overline{b}_m(a)$ the value of *b* for which $2\Pi^{pe} = 2\Pi^{dca}$ (i.e., for which industry profits are the same under pairwise exclusivity and double common agency) and with $\overline{b}_m(a)$ the value of *b* for which $\Pi^{dm} = 2\Pi^{pe}$ (i.e., for which industry profits are the same under downstream monopoly and pairwise exclusivity), the figure shows that industry profits are maximized by double common agency for $b \leq \overline{b}_m(a)$, pairwise exclusivity for $\overline{b}_m(a) \leq b \leq \overline{b}_m(a)$, and downstream monopoly for $b \geq \overline{b}_m(a)$. Industry profits under other supply networks, such as a mixed network or upstream monopoly, are always dominated by industry profits under one or more of the three networks shown in the figure.

The intuition for Figure 2 can be explained by noting that the elimination of supply links has two opposite effects on industry profits. On the one hand, it reduces the intensity of downstream competition. Specifically, a move from double common agency to pairwise exclusivity increases the degree of effective retailer differentiation, since the two retailers go from both carrying the same products to each carrying a different product, whereas a move from pairwise exclusivity to downstream monopoly completely eliminates any residual downstream competition. On the other hand, when both suppliers and retailers are differentiated, eliminating supply links causes loss of variety. As *b* increases, more intense downstream competition causes greater dissipation of industry profits and thus the former effect (softening of downstream competition) becomes progressively more important than the latter effect (loss of variety). Therefore, as *b* increases, industry profits are maximized by networks with less downstream competition and less variety.

5 Equilibria without exclusive contracts

Having characterized the supply networks that maximize industry profits, one can study the extent to which bilateral contracting between suppliers and retailers can implement these networks under different assumptions about upfront transfers and exclusive contracts. In this section, I start by considering the case in which firms can use upfront transfers but exclusive contracts are banned or not enforceable.

Proposition 2 (No exclusive contracts) Consider a bilateral duopoly model where firms cannot publicly commit to contract proposals and face the symmetric linear demand system in (8). When firms can use transfers but not exclusive contracts there exists a unique coalition-proof Nash equilibrium supply network $g_{t,ne}^*$ with

1.
$$g_{t,ne}^* = pe \text{ and transfers } t_{ne}^{pe} \in (0, \Pi^{pe} - \Pi^{um}) \text{ if and only if}$$

$$\Pi^{pe} \ge \Pi^{mix2}.$$
 (9)

2.
$$g_{t,ne}^* = dca \ and \ t_{ne}^{dca} \in \left(0, \Pi^{dca} - \Pi^{mix1}\right) \ otherwise.$$

With differentiated Bertrand competition (9) holds if and only if $b \ge \overline{b}_{t,ne}(a)$ (see shaded area in Figure 3(*a*) below). With differentiated Cournot competition (9) never holds.

Although the formal proof of Proposition 2 is quite involved, the intuition for the result is fairly straightforward. Without exclusive contracts, it is difficult (though not impossible) for supply networks with some inactive links to be self enforcing, even when these networks maximize industry profits. This is because, starting from a candidate equilibrium network with inactive links (e.g., g = dm or g = pe), under most configurations of parameters at least one supplier-retailer pair has bilateral incentives to deviate by activating one of the inactive links and, in the absence of exclusive contracts, can do so without the consent of other trading partners. As a



Figure 3: Equilibria with transfers and no exclusive contracts.

result, in equilibrium firms tend to form "too many" supply links from the point of view of industry profit maximization.

Specifically, as illustrated in Figure 3, downstream monopoly can never be supported as an equilibrium, even though it maximizes industry profits for $b > \overline{\overline{b}}_m(a)$, because, without exclusive contracts, a supplier can accept an offer from the excluded retailer, r', without having to forego the profits he earns from the other retailer, r, and can thus always be induced to do so by r'. Analogously, pairwise exclusivity cannot be supported as an equilibrium in large parts (with Bertrand competition) or the totality (with Cournot competition) of the region with $\left[\overline{b}_{m}(a),\overline{b}_{m}(a)\right]$ in which it maximizes industry profits. The relevant deviation from a can $b \in$ didate equilibrium with pairwise exclusivity is one in which a supplier, say s, opens a new link with a second retailer, say r', thus giving rise to a mixed network in which r', who now carries two products, generates vertical profits Π^{mix2} . This deviations is mutually profitable and self enforcing whenever $\Pi^{mix2} > \Pi^{pe}$, which is the case for $b < \overline{b}_{t,ne}(a)$ with Bertrand competition and always the case with Cournot competition. The difference between Bertrand and Cournot competition is explained by the fact that, for any given degree of intrinsic retailer substitutability b, the former is generally more competitive than the latter. The softening of downstream competition caused by pairwise exclusivity has, therefore, a greater positive effect on profits in the former than in the latter.

The results in Proposition 2 have been obtained under the assumption that firms can engage in *any* type of nonbinding pre-play communication, including nonbinding (but possibly self enforcing) reciprocal agreements between firms on the same side of the market not to deal with their competitor's suppliers or retailers. Since these agreements effectively amount to market allocation agreements, which are illegal in most jurisdictions even when nonbinding, one may wonder how the results would change if one prohibited deviations relying on such agreements. The following remark, which maintains the same assumptions as Proposition 2 except for the types of nonbinding agreements that are allowed, addresses this question.

Remark 1 (No exclusive contracts, no market allocation agreements) When firms cannot discuss (nonbinding) market allocation agreements, in addition to the equilibria in Proposition 2 there also exist equilibria with $g_{t,ne}^* = dca$ when condition (9) holds. This gives rise to multiple equilibria for Bertrand competition and $b \ge \overline{b}_{t,ne}(a)$ in the shaded area in Figure 3(*a*).

As one would expect, limiting the type of nonbinding communication that can take place expands the set of equilibria by reducing the ability of firms to coordinate deviations. In the specific case of the bilateral duopoly model of this section, when firms can engage in any type of communication, as was the case in Proposition 2, and condition (9) holds, the two retailers can reach a mutually profitable and self-enforcing nonbinding agreement to deviate from a candidate equilibrium with g = dca to a network g = pe in which they allocate the market by each refusing to sell a different product. This is, however, no longer possible under the restriction in Remark 1.

6 Equilibria with exclusive contracts

In this section I study the implications of allowing firms to use exclusive contracts. In principle one could adopt a general framework in which a contract between firm *i* and firm *j* can be made contingent on all other supply links in the market, i.e., on all the links that firm *i* has with other firms $h \neq j$, all the links that firm *j* has with other firms $k \neq i$, and all the links that other firms $h, k \neq i, j$ have with one another. Such contracts are, however, rarely enforceable for practical and legal reasons. Therefore, I focus on the case in which the only contingencies allowed in a contract between *i* and *j* are those requiring *i* and/or *j* to be exclusive to the other.¹⁷

¹⁷I also assume that the tariff in a nonexclusive contract between a firm *i* and a firm *j* cannot be made contingent on the volume traded by either firm with third parties $k \neq i, j$, as would instead be the case with market-share discounts or retail price parity agreements. These restraints would not make a difference in the environment with secret contracting studied in this paper (see Rey and Vergé (2016), because the fact that upstream margins are equal to zero (i.e., w = c) rules out any competition for *marginal* sales between suppliers. They would, however, have competitive effects in environments in which firms can publicly commit to contracts, as in such environments equilibrium upstream margins are generically different from zero (see Ramezzana, 2016).

Specifically, I assume that a contract between *i* and *j* can either require no exclusivity at all, or *i* to be exclusive to *j*, or *j* to be exclusive to *i*, or mutual exclusivity between *i* and *j*.

Moreover, to prevent firms from making mutually inconsistent exclusive offers, I assume that a firm *i* that offers an exclusive contract to a firm *j* is not allowed to offer contracts to other firms $k \neq j$. However, if *i* has promised exclusivity to *j*, and *j* engages in a further deviation in which it rejects *i*'s proposal, *i* is again free to submit proposals to other firms $k \neq j$ that do not belong to the deviating coalition that includes *j*.¹⁸ This last assumption, which I call the "no stranding" assumption, rules out the possibility that *j* can eliminate *i* from the market by first convincing *i* not to discuss a contract with any other firm $k \neq j$ and subsequently leaving *i* stranded without a partner (i.e., by essentially playing a "bait-and-switch" strategy).

Such bait-and-switch strategies would allow *j* to implement in a roundabout way supply networks that it could not implement directly and that are unrealistic in the unfettered contracting environment of this paper. To see this, consider the following example based on Figure 1. Assume that, starting from an equilibrium with g = dca, as in panel 1(a), a coalition {*S*1, *S*2, *R*1} reaches a nonbinding agreement to deviate to g = dm, with both suppliers agreeing to submit exclusive contract offers to R1, and consequently no contract offers to R2, as in panel 1(d). Once this nonbinding agreement has been reached, S1 and R1 may deviate further by committing to mutual exclusivity, thus excluding S2, as in panel 1(f). Absent the "no stranding" assumption introduced above, S2 would remain without a trading partner, as would R2, who was excluded in the initial deviation to g = dm. As a result, S1 and R1 would be able to implement a bilateral monopoly, an outcome they could not have achieved by deviating to mutual exclusivity directly from the candidate equilibrium with g = dca. This is an unrealistic outcome, as it is difficult to see how S1 and R1 could expect the two stranded firms, S2 and R2, not to find a way to reach an agreement to form a supply link. The "no stranding" assumption allows S2 and R2 to form such a link and, therefore, implies that a deviation to mutual exclusivity by S1 and R1 results in a pairwise exclusive supply network, as in panel 1(c), instead of resulting in a bilateral monopoly network.19

Another issue that one needs to address is the extent to which a given supply network can

¹⁸Alternatively, one can assume that when *i* promises exclusivity to *j* it can also simultaneously submit to other firms $k \neq j$ contract proposals that become valid only if *j* rejects *i*'s initial exclusive offer. This alternative assumption would yield the same results.

¹⁹The issue of how players that are excluded from a coalitional deviation will react to such a deviation has received considerable attention in the literature on coalitional equilibria and coalition formation, where it is known as the "prediction problem" (see, e.g., Bloch and Dutta (2011) for a discussion). The "no stranding" assumption adopted in the specific context of this section is reminiscent of the solution to this problem proposed by Ray and Vohra (1997), who allow players that are excluded from a deviating coalition to reorganize their strategies optimally (i.e., to play a best response).

be supported by different combinations of exclusive clauses (including, possibly, no exclusive clauses at all). Since a link between two firms can remain inactive also in the absence of contractual exclusive clauses, provided that at least one of the two firms refuses to trade with the other, exclusive clauses are not necessary to obtain any give supply network *g*. For example, the supply network with pairwise exclusivity shown in Figure 1(c) can be implemented through any type of one-way or mutual exclusive clauses between *S*1 and *R*1 and between *S*1 and *R*2 or by all firms refusing to engage in "cross trade," without any contractual exclusive clauses. As a result, the set of possible supply networks in a bilateral duopoly remains the same as in the absence of exclusive clauses is irrelevant for the equilibria of the model. On the contrary, different combinations of exclusive clauses have different implications for the feasibility or profitability of deviations from a given network, and can therefore support different networks as equilibria.

In particular, networks that are implemented with a more extensive use of exclusive clauses are generally less vulnerable to deviations in which firms attempt to expand the network by adding new supply links, because such deviations require the consent of a broader set of firms. For example, consider the supply network with pairwise exclusivity shown in Figure 1(c). If this network is supported only by a refusal by all firms to "cross trade", without any contractual exclusive clauses, a deviation that adds a link between *S*2 and *R*1 (thus implementing the mixed network in Figure 1(b)) only requires the consent of *S*2 and *R*1. If instead the same network is supported by one-way contractual exclusivity (e.g., *S*1 and *S*2 committing to be exclusive to *R*1 and *R*2, respectively, but not the other way around) the same deviation requires the consent of the coalition $\{S2, R1, R2\}$, since *R*2 must now consent to *S*2 dealing also with *R*1. Finally, if the same network is supported by mutual contractual exclusivity between *S*1 and *R*1 and between *S*1 and *R*2, the deviation requires the consent of the grand coalition $\{S1, S2, R1, R2\}$, since every firm must consent to its trading partner in the candidate equilibrium starting to deal with a new firm.

In order to study the full effects that firms can achieve by resorting to contractual exclusive clauses, I resolve the ambiguity discussed above by assuming that whenever firms *i* and *j* do not trade with each other in a candidate equilibrium network *g*, they are prevented from doing so by the most extensive combination of exclusive clauses consistent with network *g*. For example, I assume that a candidate equilibrium with the pairwise exclusive supply network in Figure 1(c) is supported by mutual exclusivity between *S*1 and *R*1 and between *S*1 and *R*2.²⁰

²⁰It is important to note that this assumption applies only to candidate equilibrium networks, not to the networks that result from deviations. Deviations are not restricted in any way and can rely on any possible

6.1 Effects of exclusive contracts on equilibrium supply networks

Under the assumptions introduced above one can establish the following result.

Proposition 3 (Exclusive contracts) In a bilateral duopoly model with the same assumptions as in Proposition 2, but in which firms can use exclusive contracts, there exist pure-strategy coalition-proof Nash equilibria with the following supply networks and transfers:

1. $g_{t,e}^* = dm$ and $t_e^{dm} = \Pi^{dm}/2$ if and only if

$$\mathbf{I}^{dm} \ge 2\Pi^{pe},\tag{10}$$

which is the case if and only if $b \ge \overline{\overline{b}}_m(a)$ (see Figure 4).

2.
$$g_{t,e}^* = pe \text{ and } t_e^{pe} \in \left[\Pi^{dm} - \Pi^{pe}, \min\left\{2\left(\Pi^{pe} - \Pi^{um}\right), \Pi^{pe}\right\}\right] \text{ if and only if (10) fails and}$$

$$2\left(\Pi^{pe} - \Pi^{um}\right) \ge \Pi^{dm} - \Pi^{pe}, \tag{11}$$

which is the case if and only if $\overline{\overline{b}}_{t,e}(a) \leq b < \overline{\overline{b}}_m(a)$ (see Figure 4).

3.
$$g_{t,e}^* = dca \text{ and } t_e^{dca} \in \left[\left(\Pi^{dm} - \Pi^{dca} \right) / 2, \Pi^{dca} - \Pi^{mix1} \right] \text{ if and only if}$$

$$2 \left(\Pi^{dca} - \Pi^{mix1} \right) \geq \Pi^{dm} - \Pi^{dca}, \qquad (12)$$

which is the case if and only if $b \leq \overline{b}_{t,e}(a)$ (see Figure 4).

There exist no other pure-strategy equilibria.

For those combinations of the supplier and retailer substitutability parameters, *a* and *b*, for which a pure-strategy equilibrium supply network exists, such a network is unique and maximizes industry profits.²¹ The latter can be seen by noting that the lines $\overline{b}_m(a)$ and $\overline{\overline{b}}_m(a)$ in Figure 4 are the same as those used in Figure 2 to illustrate the supply networks that maximize industry profits for different values of *a* and *b*. The intuition for this result is as follows.

Starting from a candidate equilibrium with some inactive supply links (e.g., $g_{t,e}^* = pe$), consider a deviation in which two firms *i* and *j* want to form a new link. In the absence of exclusive contracts, they can do so without having to ask for permission from, or forego their relationship

combination of exclusive clauses. However, deviations that use the most restrictive combination of exclusive clauses are the most likely to be self enforcing, and therefore play a prominent role in the proofs of the results that follow.

²¹Indeed, as can be seen from the proof of Proposition 3, the CPNE of the model with exclusive contracts correspond to its strong equilibria. In a model with transferable utility like the present one, when strong equilibria exist, they must always maximize the sum of the players' payoffs (i.e., of the firms' profits).



Figure 4: Equilibria with exclusive contracts and upfront transfers.

with, any other firm $k \neq i, j$. This is indeed the reason that, as shown in Section 5, equilibria with downstream monopoly or pairwise exclusivity might not be supportable without exclusive contracts even when they maximize industry profits. Things are, however, quite different if in the candidate equilibrium one or both of *i* and *j* have committed to be exclusive to other firms. For example, assume that firm *i* is part of the candidate equilibrium network and has committed to be exclusive to firm $k \neq j$. If *i* wishes to deviate by forming a supply link with *j* it must now either i) obtain consent from *k*, possibly in exchange for compensation, or ii) forego its relationship with *k*, effectively swapping *j* for *k*. Starting from equilibria with g = dm and g = pe, i) and ii) make forming a link with *j* unprofitable for *i* whenever it is unprofitable for the supply network as a whole. Specifically, starting from g = dm a supplier may wish to form a link with the excluded retailer, and starting from g = pe any supplier or retailer may wish to form a second supply link. However, exclusive contracts, through the effects in i) and ii) discussed above, make this unprofitable when g = dm and g = pe maximize industry profits.

Besides supporting equilibria with some sort of exclusivity when these maximize industry profits, exclusive contracts also eliminate (pure-strategy) equilibria with nonexclusive networks, such as g = dca, when these networks do not maximize industry profits, as is the case for $\overline{b}_m(a) \leq b < \overline{b}_{t,ne}(a)$ in Figure 3, where exclusive contracts were not available. The reason for this is that exclusive contracts tend to make deviations that rescind some links, such as deviations to g = dm or to g = pe, self-enforcing.

The latter is also the reason that pure-strategy equilibria may fail to exist for high values of a and intermediate values of b.²² For these parameter values, the availability of exclusive contracts makes both deviations in which firms exclude one of the retailers (such as a deviation to g = dm) and deviations in which they exclude one of the suppliers (such as a deviation to g = mix) become self enforcing. As preventing the first type of deviation requires large transfers t^g from retailers to suppliers (to ensure that retailers do not extract to much), whereas preventing the second type of deviations requires small transfers t^g (to ensure that suppliers do not extract too much), there may exist no t^g that can prevent all deviations.

This is, instead, not a problem for low values of *a* or extreme (low or high) values of *b*. For example, when *a* is low (i.e., suppliers are highly differentiated), a supply network with g = pe allows retailers to inherit the high degree of supplier differentiation, thus softening downstream competition without sacrificing much variety. Starting from such a network, deviations to g = dm or g = mix would not be profitable, as they would only provide a modest reduction in downstream competition and would entail a significant sacrifice in variety. Analogously, deviations from g = dca to g = dm would be very costly for low values of *b*, and deviations from g = dm to g = pe would be very costly for high values of *b* especially when *a* is also high.

As shown in the online appendix, the nonexistence of pure-strategy equilibria would not constitute a problem if one dropped the "no stranding" assumption. Specifically, dropping this assumption would make deviations from g = dca or g = pe to g = dm less likely to be self-enforcing, and would thus result in a weaker equilibrium concept, with pure-strategy equilibria existing for a broader range of parameters. This is because, absent the "no stranding" assumption, a further deviation from g = dm to mutual exclusivity by a supplier and a retailer would implement a bilateral monopoly, g = bm, which is more profitable than the pairwise exclusive network, g = pe, that the same deviation would instead implement with the "no stranding" assumption. As shown in the online appendix, without the "no stranding" assumption the model would have multiple pure-strategy equilibria for high values of *a* and intermediate values of *b*, instead of having no pure-strategy equilibria. Regardless of its consequences for the existence and uniqueness of equilibria, as explained above, I believe that the "no stranding" assumption

²²As noted by Bernheim, Peleg and Whinston (1987), the existence coalition-proof Nash equilibria, even in mixed strategies, cannot be guaranteed for general classes of games. Specifically, finding sufficient conditions for the existence of such equilibria is far from obvious, unless the game played by any subset of players, given any set of actions of the remaining players, has a unique Nash equilibrium. Since this uniqueness condition does not necessarily apply to the bilateral oligopoly game presented here, one cannot assert with certainty that there exist mixed-strategy equilibria in the parameter region where pure-strategy equilibria do not exist, although this may be the case. Given the complexity of the analysis, I have not attempted a characterization of mixed-strategy equilibria.

(or some other assumption with a similar function) is necessary for a realistic analysis of the long-term structure of supply networks.

6.2 Effects of exclusive contracts on welfare and firms' profits

The results obtained above have the following welfare implications.

Proposition 4 (Welfare) Whenever exclusive contracts are adopted and affect the equilibrium structure of supply networks they reduce consumer and overall welfare.

The result in Proposition 4 is straightforward (a proof relying on the utility function underlying the inverse linear demand in (8) is provided in the online appendix enclosed with this submission). Specifically, with Bertrand downstream competition, exclusive contracts cause the equilibrium network to switch from double common agency to pairwise exclusivity for $\overline{b}_{t,e}(a) \leq b < \overline{b}_{t,ne}(a)$ and from pairwise exclusivity to downstream monopoly for $b \geq \overline{b}_m(a)$. With Cournot downstream competition, instead, they cause the equilibrium network to switch from double common agency to pairwise exclusivity for $\overline{b}_m(a) \leq b < \overline{b}_m(a)$ and from double common agency to downstream monopoly for $b \geq \overline{b}_m(a)$. In all these cases they soften downstream competition, thus leading to higher prices, and reduce the variety of supplier retailer combinations available in the market. Both effects unambiguously reduce consumer and overall welfare.²³

Less straightforward are, instead, the effects of the availability of exclusive contracts on the equilibrium profits earned by individual suppliers and individual retailers. Although, as explained in Section 3.2, the simultaneous contracting model used in this paper does not yield exact predictions regarding the equilibrium level of transfers, it nevertheless provides ranges within which such transfers, and thus the equilibrium profits of individual suppliers and retailers, must lie. If a change in the environment, such as the availability of exclusive contracts, causes these ranges to shift entirely to the right or the left of their initial position, the model adopted in this paper is sufficient to determine the distributional effects of that change. This is the approach used in deriving the following result.

Proposition 5 (Distribution of profits) When exclusive contracts become available and are adopted in equilibrium, so that the resulting equilibrium supply network is $g_{t,e}^* = pe$ or $g_{t,e}^* = dm$, they make

²³As is well known, under certain conditions (e.g. when the seller and/or the buyer can make relationshipspecific investments) exclusive contracts can enhance welfare by aligning investment incentives. In order to focus on the competitive effects of exclusive contracts, these potential efficiencies are not addressed in this paper.

suppliers strictly better off and retailers strictly worse off. When exclusive contracts become available but are not adopted in equilibrium, so that the equilibrium supply network remains $g_{t,e}^* = dca$, they make suppliers no worse off (and possibly strictly better off) and retailers no better off (and possibly strictly worse off).

The intuition for this results can be understood by noting that, besides possibly affecting the level of equilibrium industry profits, the availability of exclusive contracts may also affect the shares of any given amount of profits that suppliers and retailers are able to extract. Specifically, the upper bound on the profits that the firms on one side of the market can extract in a given equilibrium are determined by the extent to which the firms on the opposite side of the market can profitably drop them from their supply network in deviations that are self enforcing. In other words, the profits of the firms on one side of the market are determined by the credible disagreement payoffs available to the firms on the opposite side of the market. The availability of exclusive contracts makes a broader range of deviations self enforcing relative to an environment without exclusive contracts. For example, deviations to g = um, in which a supplier is excluded, or to g = dm, in which a retailer is excluded, can be self enforcing only if exclusive contracts are available. As shown in the proof of Proposition 3, whereas deviations to g = um, which limit the bargaining power of suppliers, are not sufficiently profitable to be self-enforcing (i.e., credible), deviations to g = dm, which limit the bargaining power of retailers, are generally sufficiently profitable to be self enforcing. Loosely speaking, this is the case because deviations to g = um sacrifice product variety without softening downstream competition, whereas deviations to g = dm, though also costly in terms of variety, can increase profits by eliminating downstream competition. As a result, by making deviations to g = dm self enforcing, the availability of exclusive contracts improves the credible disagreement payoffs of suppliers and shifts the balance of power in their favor.

Note that, when exclusive contracts are allowed, retailers have no way of preventing this outcome. Each individual retailer would have incentives to accept contracts that made her a downstream monopolist if she were offered such contracts, and (at least in this model) retailers cannot credibly commit not to accept such contracts. One would, however, expect that if retailers were given a say in public policy towards exclusive contracts before the contracting game is played, they would generally oppose the availability of such contracts.

The mechanism by which the availability of exclusive contracts can affect the distribution of profits in the present model is different from the mechanism discussed in O'Brien and Shaffer (1997) and at work in Liebman (2016) and Ho and Lee (2017). In those papers, the firms on one

side of the market (typically downstream firms) can extract higher profits by (credibly) committing ex-ante to accept only a limited number of offers and honoring that commitment by actually excluding some firms in equilibrium. The mechanism in this paper does not rely on ex-ante commitment to exclude some firms in equilibrium, but rather on the effects of exclusive contracts on out-of-equilibrium alternatives. It is, therefore, closer in spirit to that studied by Bernheim and Whinston (1998) in a different setting with entry deterrence and without downstream competition.

7 Ex-post bargaining and hold up

To understand better the role that upfront transfers play in shaping supply networks, it is helpful to compare the equilibria characterized in Sections 5 and 6 with those of a modified game in which firms cannot use transfers at the network formation stage and must instead bargain bilaterally under some degree of hold-up only after a supply network has been formed.²⁴ This modified game develops in three stages. In stage 1, firms form supply links by playing Myerson's (1991) network formation game. Specifically, all firms simultaneously announce a list of firms on the other side of the market with which they are willing to deal, and a supply link between two firms is formed if and only if each firm has indicated the other as a partner. In Subsection 7.1 I assume that firms cannot commit to exclusivity as part of their announcements, whereas in Subsection 7.2 I allow for such commitment. No transfers are possible at this stage. In stage 2, all pairs of firms with a supply link bargain bilaterally, simultaneously and secretly over nonlinear supply contracts with fixed fee F_{sr} and wholesale price w_{sr} , splitting the surplus according to the generalized Nash bargaining solution, with a share β going to the supplier and a share $(1 - \beta)$ to the retailer. Importantly, bilateral bargaining can take place only between firms with an existing link and no new links can be formed at this stage, which exposes firms to holdup in negotiations. Finally, in stage 3, after having observed which supply links resulted from stage 1 (i.e., the prevailing supply network) but not the specific terms (e.g., the wholesale prices) that resulted from contracting in stage 2, retailers compete in the downstream market.

This game is similar to that studied by Rey and Vergé (2016) and is analyzed here mostly to provide context for the results obtained in previous sections, rather than as an original contribution in itself. This notwithstanding, the version of the model presented in this section features

²⁴As already discussed, complete long-term contracts would play a role similar to upfront transfers, because they would allow firms to commit to leave a certain level of profits to their future trading partners at the moment in which supply networks are formed. The analysis of contracting under hold-up presented in this section should therefore be interpreted as referring to environments in which both transfers and complete long-term contracts are unavailable or severely limited.

some differences and additional contributions relative to Rey and Vergé (2016). Specifically, I assume that retailers observe the full set of active supply links (i.e., the prevailing supply network) before competing in the downstream market,²⁵ and also study the cases, not covered by Rey and Vergé, in which firms can adopt exclusive contracts and in which downstream competition is Cournot.

Given the gross vertical profits generated by downstream competition in stage 3, which remain the same as in Sections 4 through 6 (and are derived in the online appendix for the case of linear demand), one can solve for the contracts that result from bilateral negotiation in stage 2. As in previous sections, all wholesale prices are equal to marginal cost. The profits earned by a supplier *s* in supply network *g* correspond, therefore, to the sum of fixed fees, $\sum_{r \in R} \ell_{sr} F_{sr}^g$ charged by *s* in that network. For each network *g* in which $\ell_{sr} = 1$ the fixed fee F_{sr}^g is determined according to the generalized Nash bargaining solution by solving the following equation for F_{sr}^g , with $g \setminus sr$ denoting the supply network obtained from *g* when *s* and *r* do not form a supply link,

$$F_{sr}^{g} + \sum_{j \neq r} \ell_{sj} F_{sj}^{g} = \sum_{j \neq r} \ell_{sj} F_{sj}^{g} + \beta \left[\left(\sum_{j \neq r} \ell_{sj} F_{sj}^{g} + \Pi_{r}^{g} - \sum_{i \neq s} \ell_{ir} F_{ir}^{g} \right) - \left(\sum_{j \neq r} \ell_{sj} F_{sj}^{g} + \Pi_{r}^{g \setminus sr} - \sum_{i \neq s} \ell_{ir} F_{ir}^{g} \right) \right],$$
which reduces to

which reduces to

$$F_{sr}^{g} = \beta \left(\Pi_{r}^{g} - \Pi_{r}^{g \setminus sr} \right).$$
(13)

Using (13) one can calculate the following supplier (left column) and retailer (right column) profits for each possible supply network in the bilateral duopoly model introduced in Section 4.

$$\begin{aligned} \pi_{s}^{dca} &= 2\beta \left(\Pi^{dca} - \Pi^{mix1} \right), & \pi_{r}^{dca} &= \Pi^{dca} - 2\beta \left(\Pi^{dca} - \Pi^{mix1} \right), \\ \pi_{s}^{pe} &= \beta \Pi^{pe}, & \pi_{r}^{pe} &= (1 - \beta) \Pi^{pe}, \\ \pi_{s}^{dm} &= \beta \left(\Pi^{dm} - \Pi^{bm} \right), & \pi_{r}^{dm} &= \Pi^{dm} - 2\beta \left(\Pi^{dm} - \Pi^{bm} \right), \\ \pi_{s}^{um} &= \beta \Pi^{um}, & \pi_{r}^{um} &= (1 - \beta) \Pi^{um}, \\ \pi_{s}^{bm} &= \beta \Pi^{bm}, & \pi_{r}^{bm} &= (1 - \beta) \Pi^{bm}, \\ \pi_{s}^{mix1} &= \beta \left(\Pi^{mix2} - \Pi^{um} \right), & \pi_{r}^{mix1} &= (1 - \beta) \Pi^{mix1}, \\ \pi_{s}^{mix2} &= \beta \left(\Pi^{mix1} + \Pi^{mix2} - \Pi^{pe} \right), & \pi_{r}^{mix2} &= \Pi^{mix2} - \beta \left(2\Pi^{mix2} - \Pi^{pe} - \Pi^{um} \right). \end{aligned}$$

²⁵Rey and Vergé (2016) assume that retailers do not observe the prevailing downstream market structure before setting retail prices and, following an unobserved deviation involving one of their rivals, behave as if the market structure was the one prescribed by the candidate equilibrium of the supply network formation game.

The profits in (14) constitute the payoffs of the link formation game played by firms in stage 1, which I turn to study next, first for the case in which commitment to exclusivity at the network formation stage is not possible and then the case in which it is.²⁶

7.1 No ex-ante commitment to exclusivity

The main difference between the contracting game with upfront transfers studied in Section 5 and the game with ex-post bargaining studied in this subsection lies in the ability of firms to induce other firms to participate in deviations *in which new supply links are formed*. Specifically, when firms can use upfront transfers as in Section 5 and a coalition of firms would find it jointly profitable to enter into new supply contracts, there always exists a profile of upfront transfers that makes it individually profitable for each firm in the coalition to enter into those contracts. In the game with ex-post bargaining studied in this section, instead, the division of the joint profits resulting from the formation of a new supply link is determined by the firms' relative ex-post bargaining power and may not make each firm better off even if forming the new link is jointly profitable.

In the absence of exclusive contracts, upfront transfers do not play any role in facilitating deviations *in which firms rescind one or more supply links*. Without exclusive contracts, any agreement between a supplier *s* and a retailer *r* that, say, supplier *s* will not trade with another retailer $r' \neq r$ is nonbinding and must therefore be self enforcing. This implies that any transfer from *r* to *s* in exchange for a nonbinding promise by *s* to be exclusive with *r* is either unnecessary (when the promise is self-enforcing and thus *s* would have complied even without the transfer) or ineffective (when the promise is not-self enforcing).

The fact that upfront transfers facilitate agreements to create new links and, in the absence of exclusive contracts, do not have any effect on agreements to rescind existing links suggests that, at least for some parameter regions, equilibria with upfront transfers should tend to have more supply links than equilibria without upfront transfers. This intuition is confirmed by the following result.

Proposition 6 (No upfront transfers, no commitment to exclusivity) *Consider the same bilateral duopoly model as in Proposition 2 but assume that firms cannot use upfront transfers at the network*

²⁶In this game, firms may have incentives to commit to exclusivity at the time of forming the supply network in stage 1, not at the time of engaging in ex-post bilateral bargaining over supply contracts in stage 2. Once firms reach stage 2, commitment to exclusivity is of no value because, by assumption, firms would not be able to form new links anyhow, and no firm would find it profitable to reduce its outside options by contractually renouncing some of the links that it has already formed. Therefore, the exclusivity studied in this subsection is better thought of as being achieved through some sort of ex-ante technological commitment in stage 1, rather than through contractual clauses in stage 2.

formation stage. There always exists a unique coalition-proof Nash equilibrium supply network $g_{nt,ne}^*$ with

1. $g_{t,ne}^* = pe$ if and only if

$$2\Pi^{pe} > \Pi^{mix1} + \Pi^{mix2}.$$
(15)

2. $g_{nt,ne}^* = dca$ otherwise.

Condition (15) holds for $b \ge \overline{b}_{nt,ne}(a)$, where $\overline{b}_{nt,ne}(a) < \overline{b}_{t,ne}(a)$ is shown in Figure 5.



Figure 5: Equilibria without transfers (no exclusive contracts).

Figure 5 compares the equilibrium supply networks in the game with ex-post bargaining studied in this section to the equilibrium supply networks in the game with upfront transfers characterized in Section 5 and to the supply networks that maximize industry profits characterized in Section 4. As in a model with upfront transfers, in the absence of exclusive contracts downstream monopoly is not an equilibrium in a model with ex-post bargaining either, even though it maximizes industry profits for $b \ge \overline{b}_m(a)$.²⁷ In particular, although the absence of upfront transfers hinders to some extent a deviation from downstream monopoly in which the excluded retailer forms a new link with one (or more) of the suppliers for the reasons discussed above, such deviation continues to be mutually profitable also with ex-post bargaining.

²⁷To minimize clutter, I did not include the region in which downstream monopoly maximizes industry profits in Figure 5.

Ex-post bargaining does, instead, make a difference for the regions in which pairwise exclusivity and double common agency can be supported as equilibria. Specifically, with ex-post bargaining the region of parameters for which pairwise exclusivity arises in equilibrium expands, with the lower bound on *b* in Figure 5 shifting downward from $\overline{b}_{t,ne}(a)$ to $\overline{b}_{nt,ne}(a)$ (where $\overline{b}_{t,ne}(a) = 1$ for Cournot competition) and pairwise exclusivity replacing double common agency as the unique equilibrium outcome in the shaded regions.

The intuition for this is closely related to the discussion that precedes Proposition 6. Specifically, just as in the case with upfront transfers, the relevant self-enforcing deviation from a candidate equilibrium with pairwise exclusivity involves a supplier s starting to trade with a second retailer r' (or vice versa), thus implementing a mixed network. Such a deviation yields incremental joint profits of $(\Pi^{mix2} - \Pi^{pe})$ for s and r'. With upfront transfers, s and r' can find a way to profit individually from this deviation provided that $\Pi^{mix^2} > \Pi^{pe}$ (see Proposition 2). With ex-post bargaining, instead, the formation of the new link between s and r' also affects the bargaining power of these two firms in their negotiations over other links and thus the (inframarginal) profits that they can extract from those other links. The result is that both parties profit from a deviation from pairwise exclusivity to a mixed network only if $\Pi^{mix1} + \Pi^{mix2} > 2\Pi^{pe}$, which is more stringent than $\Pi^{mix^2} > \Pi^{pe}$. The region of parameters where pairwise exclusivity is an equilibrium is therefore larger relative to that for a model with upfront transfers. This is accompanied by a corresponding reduction of the region where double common agency is an equilibrium, due mainly to that fact that deviations that move the market away from double common agency by rescinding links are more self-enforcing, though not necessarily more jointly profitable, when upfront transfers are not available.

Finally, ex-post bargaining affects overall industry profits differently in different parameter regions. As discussed above, ex-post bargaining shifts the equilibrium from double common agency to pairwise exclusivity in the entire shaded region in Figure 5, i.e., for $b \in [\overline{b}_{nt,ne}(a), \overline{b}_{t,ne}(a)]$. This shift increases total industry profits in the region without hashing in which $b \in [\overline{b}_m(a), \overline{b}_{t,ne}(a)]$ and reduces them in the hashed region in which $b \in [\overline{b}_{nt,ne}(a), \overline{b}_m(a)]$. Intuitively, the fact that ex-post bargaining makes it harder to form new links has two opposite effects on overall industry profits. On the one hand, it makes it harder for supplier-retailer pairs to impose negative externalities on other firms by forming new links that reduce the profits of those firms (this positive effect dominates in the shaded region without hashing). On the other hand, it also makes it harder for firms to realize gains from trade when these exist (this negative effect dominates in the hashed region). The fact that upfront transfers that cannot be made contingent on exclusivity do not necessarily lead to the maximization of total payoffs in games of

network formation with externalities has also been stressed in a different context by Bloch and Jackson (2007).²⁸

7.2 Ex-ante commitment to exclusivity

Assume now that, when firms announce which other firms they are willing to form a link with at the network formation stage, they can make that announcement contingent on the different type of exclusivity studied in Section 6. For example, a supplier s can announce that he is willing to form a link with a retailer r only if he obtains exclusivity from r or, alternatively, that he is willing to offer exclusivity to r. As above, once the supply network has been formed, firms that are connected by a supply link proceed to bargain bilaterally under some degree of hold-up. The following proposition characterizes the equilibria that arise in this environment.

Proposition 7 (No upfront transfers, with commitment to exclusivity) In a bilateral duopoly model where firms can commit to exclusivity but not upfront transfers at the network formation stage, there exist coalition-proof Nash equilibria with

1. $g_{nt,e}^* = pe$ if and only if

$$\Pi^{pe} \ge 2\left(\Pi^{dca} - \Pi^{mix1}\right). \tag{16}$$

which is the case if and only if $b > \overline{b}_{nt,e}(a)$ in Figure 6 below.

2. $g_{nt,e}^* = dca and g_{nt,e}^* = mix if and only if$

$$\Pi^{mix1} + \Pi^{mix2} \ge 2\Pi^{pe},\tag{17}$$

which is the case if and only if $b \leq \overline{\overline{b}}_{nt,e}(a) = \overline{b}_{nt,ne}(a)$ in Figure 6 below, with $\overline{\overline{b}}_{nt,e}(a) > \overline{b}_{nt,e}(a)$.

The availability of exclusive contracts and the absence of upfront transfers make it more difficult for firms to deviate profitably by forming new supply links. Exclusive contracts tend, therefore, to expand the region of parameters where networks with some type of exclusivity, such as a mixed network or pairwise exclusivity, can be supported as equilibria. Specifically, in addition to the equilibrium with double common agency that exists also in the absence of

²⁸Elliott (2015) also explores the effects of upfront transfers (relative to ex-post bargaining under hold up) on the efficiency of networks. The environment and issues that he studies are, however, quite different from those in this paper. Specifically, he considers markets without externalities (i.e., in which downstream firms do not compete with one another) and focuses on the firms' incentives to undertake relation-specific investments. In such an environment transfers also have two opposite effects on overall profits: they alleviate underinvestment by solving the hold-up problem but can encourage firms to overinvest in new links in order to boost their outside options.



Figure 6: Equilibria with exclusive contracts and no transfers.

exclusive contracts for $b < \overline{\overline{b}}_{nt,e}(a)$, exclusive contracts support an equilibrium with a mixed network for $b < \overline{b}_{nt,e}(a)$ and equilibria with a mixed network and pairwise exclusivity for $b \in [\overline{b}_{nt,e}(a), \overline{\overline{b}}_{nt,e}(a)]$ in Figure 6.

One may wonder why there exist equilibria with exclusivity even when b = 0 in Figure 6, i.e., even when there is no downstream competition between retailers and thus exclusivity always reduces industry profits by reducing variety. The reason for this is that some firms have more ex-post bargaining power under certain supply networks with exclusivity than under alternative networks without exclusivity. For example, suppliers have more ex-post bargaining power under pairwise exclusivity, in which retailers have no ex-post alternative to their only supplier, than under double common agency, in which retailers can shift ex-post some of their business to the other supplier in case of disagreement during bilateral bargaining. In the absence of upfront transfers, the firms that would see their bargaining power fall as a consequence of deviations to networks with less exclusivity cannot be compensated for their loss and would thus block such deviations.

8 Conclusion

This paper has studied the formation of supply networks in environments with upstream and downstream competition in which firms can use upfront transfers and exclusive contracts. I have shown that, when contracts are secret, all (coalition-proof) equilibria are characterized by marginal input prices equal to marginal cost. Moreover, when exclusive contracts are not available and retailers are sufficiently close substitutes, equilibrium supply networks tend to have more supply links and more intense downstream competition than the networks that maximize industry profits. Exclusive contracts make it easier to support equilibrium networks with fewer supply links, thus eliminating the divergence between equilibrium and industry-profit maximizing networks and harming welfare through lower variety and higher prices. Finally, if the division of profits must take place through ex-post bargaining, for example because firms must make specific investments before starting to negotiate, it is more difficult for firms to organize mutually profitable deviations to broader networks and is thus easier to support equilibria with fewer supply links relative to an environment with upfront transfers or long-term contracts.

The model presented in this paper has been developed for the main purpose of studying the determinants and welfare effects of different supply networks, not of precisely predicting the division of profits between suppliers and retailers. As a result, it only characterizes lower and upper bounds on the transfers that support different equilibrium supply networks. This approach is sufficient to determine unambiguously the qualitative effects of certain changes in the environment on the distribution of profits. For example, it predicts that the availability of exclusive contracts unambiguously favors suppliers over retailers, by allowing the former to threaten credibly to exclude the latter. To obtain more precise predictions on the exact division of profits one would, however, need to adopt a specific coalitional bargaining game at the network-formation stage. Such an extension would add further complexity to an already complex environment and would be unlikely to affect the equilibrium structure of networks, unless the specific coalitional bargaining protocol adopted i) limits the firms' ability to continue making one another proposals until all possibilities have been explored, or ii) allows firms to credibly commit ex-ante (i.e., before starting negotiations) to restrict the number of counterparts with which they are willing to contract in order to elicit better offers from the other side of the market as in, e.g., Liebman (2016). Further work on this aspect may, however, prove useful.

Finally, this paper has focused on the case in which suppliers cannot publicly commit to marginal input prices. However, in some industries suppliers can attain some form of public commitment through various means.²⁹ This commitment can affect the structure of equilibrium

²⁹Examples of regulatory obligations include the healthcare industry, in which some U.S. states (e.g. New Hampshire) have recently started to collect and publish the bilateral reimbursement rates agreed for every procedure by healthcare providers and insurers; the alcohol distribution industry, in which some U.S. license states require wholesalers to post and commit to wholesale prices for 30 days; and the automobile industry, in which the franchise laws of many states require manufacturers to offer the same price to all dealers and

supply networks because its potential to soften downstream competition and increase industry profits is greater in certain networks than in others. Moreover, since it typically yields wholesale prices that are different from the marginal cost of production, it introduces externalities between suppliers, in addition to the externalities between retailers that are also present in the environment without public commitment of this paper. I am studying these and other implications of public commitment in ongoing work.

References

- Bernheim, Douglas, Bezalel Peleg and Michael Whinston (1987), "Coalition-Proof Nash Equilibria I: Concepts," *Journal of Economic Theory*, Vol. 42, pp. 1-12.
- Bernheim, Douglas and Michael Whinston (1998), "Exclusive Dealing," Journal of Political Economy, Vol. 106 No. 1, pp. 64-103.
- **Besanko, David** and **Martin Perry** (1994), "Exclusive Dealing in a Spatial Model of Retail Competition," *International Journal of Industrial Organization*, Vol. 12, pp. 297-329.
- **Bloch, Francis** and **Bhaskar Dutta** (2011), "Formation of Networks and Coalitions," in J.Benhabib, A.Bisin and M. Jackson (eds.), *Handbook of Social Economics*, Vol 1, pp. 729-779, North Holland.
- **Bloch, Francis** and **Matthew Jackson** (2007), "The Formation of Networks with Transfers among Players," *Journal of Economic Theory*, Vol. 133(1), pp. 83-110.
- **Chatterjee, Kalyan, Bhaskar Dutta, Debraj Ray** and **Kunal Sengupta** (1993), "A Noncooperative Theory of Coalitional Bargaining," *Review of Economic Studies*, Vol. 60(2), pp. 463-477.
- **Collard-Wexler, Allan, Gautam Gowrisankaran** and **Robin Lee** (2017), "'Nash-in-Nash' Bargaining: A Microfoundation for Applied Work," forthcoming, *Journal of Political Economy*.
- **Crawford, Gregory** and **Ali Yurukoglu** (2012), "The Welfare Effects of Bundling in Multichannel Television Markets," *American Economic Review*, Vol. 102, No. 2, pp. 643-685.
- **Crémer, Jacques** and **Michael Riordan** (1987), "On Governing Multilateral Transactions with Bilateral Contracts," *RAND Journal of Economics*, Vol. 18, pp. 436-451.
- **Dobson, Paul** and **David Waterson** (1996), "Exclusive Trading Contracts in Successive Differentiated Duopoly," *Southern Economic Journal*, Vol. 63 No. 2, pp. 361-377.
- Elliott, Matthew (2015), "Inefficiencies in Networked Markets," American Economic Journal: Microeconomics, Vol. 7(4), pp. 43-82.

prohibit discounts and rebates. For a more detailed discussion see Ramezzana (2016).

- Gowrisankaran, Gautam, Aviv Nevo, and Robert Town (2015), "Mergers When Prices Are Negotiated: Evidence from the Hospital Industry," *American Economic Review*, Vol. 105(1), pp. 172-203.
- Hermalin, Benjamin and Michael Katz (2013), "Product Differentiation through Exclusivity: Is there a One-Market-Power-Rent Theorem?", *Journal of Economics and Management Strategy*, Vol. 22, pp. 1-27.
- **Horn, Henrick** and **Asher Wolinsky** (1988), "Bilateral Monopolies and Incentives for Merger," *RAND Journal of Economics*, Vol. 19(3), pp. 408-419.
- **Ho, Kate** and **Robin Lee** (2017), "Equilibrium Provider Networks: Bargaining and Exclusion in Health Care Markets," Unpublished.
- Jackson, Matthew (2008), Social and Economic Networks, Princeton, NJ: Princeton University Press.
- Kranton, Rachel and Deborah Minehart (2001), "A Theory of Buyer-Seller Networks," *American Economic Review*, Vol. 91(3), pp. 485-508.
- Lee, Robin and Kyna Fong (2013), "Markov-Perfect Network Formation: An Applied Framework for Bilateral Oligopoly and Bargaining in Buyer-Seller Networks," Unpublished.
- Liebman, Eli (2016), "Bargaining in Markets with Exclusion: An Analysis of Health Insurance Networks," Unpublished.
- Maskin, Eric (2003), "Bargaining, Coalitions and Externalities," Presidential Address to the Econometric Society.
- McAfee, Preston and Marius Schwartz (1994), " Opportunism in Multilateral Vertical Contracting: Nondiscrimination, Exclusivity, and Uniformity," *American Economic Review*, Vol. 84, pp. 210-230.
- **Myerson, Roger** (1991), *Game Theory: Analysis of Conflict*, Cambridge, MA: Harvard University Press.
- **Nocke, Volker** and **Patrick Rey** (2016), " Exclusive Dealing and Vertical Integration in Interlocking Relationships," Unpublished.
- **O'Brien, Daniel** and **Greg Shaffer** (1992), "Vertical Control with Bilateral Contracts," *RAND Journal of Economics*, Vol. 23 (Autumn), pp. 299-308.
- **O'Brien, Daniel** and **Greg Shaffer** (1993), "On the Dampening-of-Competition Effect of Exclusive Dealing," *Journal of Industrial Economics*, Vol. XLI No. 2 (June), pp. 215-221.
- **O'Brien, Daniel** and **Greg Shaffer** (1997), "Nonlinear Supply Contracts, Exclusive Dealing, and Equilibrium Market Foreclosure," *Journal of Economics and Management Strategy*, Vol. 6 No. 4,

pp. 755-785.

- OECD (2013), "Competition Issues in Television and Broadcasting," DAF/COMP/GF(2013)13.
- **Ostrovsky, Michael** (2008), "Stability in Supply Chain Networks," *American Economic Review*, Vol. 98(3), pp. 897-923.
- **Prat, Andrea** and **Aldo Rustichini** (2003), "Games Played Through Agents," *Econometrica*, Vol. 71 No. 4, pp. 989-1026.
- **Ramezzana**, **Paolo** (2016), "Nonlinear Contracts and Vertical Restraints in Bilateral Duopoly," Unpublished.
- Ray, Debraj and Rajiv Vohra (1997), "Equilibrium Binding Agreements," Journal of Economic Theory, Vol. 73, pp. 30-78.
- **Ray, Debraj** and **Rajiv Vohra** (1999), "A Theory of Endogenous Coalition Structures," *Games and Economic Behavior*, Vol. 26, pp. 286-336.
- **Rey, Patrick** and **Thibaud Vergé** (2004), "Bilateral Control with Vertical Contracts", *RAND Journal of Economics*, Vol. 35, pp. 728-746.
- **Rey, Patrick** and **Thibaud Vergé** (2016), "Secret Contracting in Multilateral Relations," TSE Working Paper, No. 16-744, December. 728-746.
- **Shapley, Lloyd** (1953), "A Value for n-Person Games," in H. Kuhn and A. Tucker (eds.), Contributions to the Theory of Games, II, Princeton: Princeton University Press, 307-317.
- **Sinkinson, Michael** (2014), "Pricing and Entry Incentives with Exclusive Contracts: Evidence from Smartphones," Unpublished.
- Weeds, Helen (2016), "TV Wars: Exclusive Content and Platform Competition in Pay TV," *Economic Journal*, Vol. 126, Issue 594, pp.1600-1633.

APPENDIX A

Proofs of Lemmas and Propositions

Proof of Proposition 1 – I first show that, for any network *g*, any profile of self-enforcing wholesale prices must be equal to marginal cost, i.e., w(g) = c. I then argue that this implies that there does not exist any self-enforcing (bilateral or multilateral) deviation from w(g) = c and therefore that there always exist a unique CPNE profile of wholesale prices with w(g) = c.

A CPNE profile of wholesale prices must be immune to jointly profitable bilateral deviations by any supplier-retailer pair *sr*. Such bilateral deviations are always self-enforcing when jointly profitable because, in a CPNE, further deviations can only be undertaken by *s* alone or *r* alone and cannot modify the wholesale prices in the contracts with other firms. Therefore, if in their initial bilateral deviation *s* and *r* agreed on a contract $\langle t_{sr}, w_{sr} \rangle$ that left both of them better off given the strategies of all other players, neither *s* nor *r* can improve on that contract by deviating unilaterally.

The above implies that a CPNE profile of wholesale prices is always a contract equilibrium (see O'Brien and Shaffer, 1992), i.e., it maximizes the joint profits of each supplier-retailer pair, taking as given the contracts entered into by other pairs. In any contract equilibrium, and thus in any CPNE, it must be $w_{sr}(g) = c$ for all $s \in S$ and $r \in R$ with $\ell_{sr} = 1$. This can be demonstrated as follows. Consider a deviation by a coalition that involves only *s* and *r*. Given the assumptions of the model and the equilibrium concept adopted, the only wholesale price that can be changed by this bilateral deviation is w_{sr} . This is the case because, given the wholesale price proposals $w_{r'}^s$ by any other retailer $r' \neq r$, s cannot unilaterally change the wholesale price $w_{sr'}$ in his contract with r'. Moreover, since r' is not part of the original deviating coalition $\{s, r\}$, and CPNE restricts further deviations to subcoalitions, s cannot orchestrate further deviations to persuade r' to change her proposal $w_{r'}^s$. The fact that in this deviation s cannot change wholesale prices other than w_{sr} , and the fact that this is common knowledge, makes it unnecessary to specify beliefs for r in this deviation. Finally, since the change in w_{sr} is unobserved to other retailers, it does not affect the quantities (with Cournot competition) or prices (with Bertrand competition) chosen by those retailers. A small change in w_{sr} has, therefore, the following effect on the joint profits $\pi_{sr} = \pi_s + \pi_r$ of *s* and *r*,

$$\frac{d\pi_{sr}}{dw_{sr}} = \sum_{i \in S_r} \frac{\partial\pi_r}{\partial x_{ir}} \frac{dx_{ir}}{dw_{sr}} + g_{sr} + \sum_{j \in R_s} \left(w_{sj} - c \right) \frac{dq_{jr}}{dw_{sr}} - g_{sr}, \tag{A-1}$$

where x_{ir} represents the choice variable of retailer r and is equal to q_{ir} for Cournot competition and p_{ir} for Bertrand competition, and S_r is the set of suppliers with which retailer r has a link (i.e., $s \in S_r$ if $\ell_{sr} = 1$) and R_s is the set of retailers with which supplier s has a link (i.e., $r \in R_s$ if $\ell_{sr} = 1$). In a contract equilibrium, and thus in a CPNE, the wholesale price w_{sr} must maximize the joint profits of any sr pair with $\ell_{sr} = 1$, otherwise there would exist a profitable and self-enforcing bilateral deviation for at least one of these pairs. It must therefore be $d\pi_{sr}/dw_{sr} = 0$ for all s and r. Moreover, profit maximization by retailer r implies that $\partial \pi_r / \partial x_{ir} = 0$ for all $i \in S_r$, therefore the first term in (A-1) is always equal to zero by the envelope theorem. This implies that the following $R_s \times R_s$ system of equations must hold in a CPNE for all $s \in S_r$

$$\frac{d\pi_{sr}}{dw_{sr}} = \sum_{j \in R_s} \left(w_{sj} - c \right) \frac{dq_{sj}}{dw_{sr}} = 0, \text{ for all } r \in R_s.$$
(A-2)

If downstream competition is Cournot, $dq_{sj}/dw_{sr} = 0$ for all $j \neq r$ and, for all $s \in S_r$, (A-2) reduces to

$$(w_{sr}-c)\frac{dq_{sr}}{dw_{sr}}=0, \text{ for all } r \in R_s.$$
(A-3)

Since $dq_{sr}/dw_{sr} < 0$ this implies $w_{sr} = c$ for all $s \in S_r$ and $r \in R_s$.

If downstream competition is instead Bertrand, $dq_{sj}/dw_{sr} \ge 0$ for $j \ne r$. In particular, although $dp_{ij}/dw_{sr} = 0$ for $j \ne r$, one still has $dp_{ir}/dw_{sr} \ne 0$ for all $i \in S_r$, and changes in the prices p_{ir} affect all quantities, including q_{sj} for $j \ne r$. The relevant conditions remain therefore those given in (A-2). These conditions can be written in matrix form as $(\mathbf{w}_s - c) \mathbf{M} = \mathbf{0}$, where \mathbf{w}_s is a $1 \times R_s$ vector, and \mathbf{M} is an $R_s \times R_s$ matrix with the expression in (A-2) constituting the typical element for column j and row r. Given Assumption 1, \mathbf{M} has a dominant diagonal, i.e., $|dq_{sr}/dw_{sr}| > \sum_{j \ne r} |dq_{sj}/dw_{sr}|$ for all $r \in R_s$. By the Levy-Desplaques theorem \mathbf{M} is therefore invertible and the unique solution to (A-2) is $w_{sr} = c$ for all $r \in R_s$.

The results above can be used to establish that w(g) = c is not only a necessary, but also a sufficient, condition for w(g) to constitute a CPNE profile of wholesale prices in a network g. Even if there existed a mutually profitable multilateral deviation that set the wholesale prices of two or more retailers at a level different from marginal cost (as might, for example, be profitable in the case of Bertrand competition and highly substitutable retailers), such a deviation would not be self-enforcing, because there would always exist further profitable and self-enforcing bilateral deviations to marginal cost pricing for at least one supplier-retailer pair.

Proof of Lemma 1 – I first prove necessity and then sufficiency. Consider a network *g* and a mutually profitable deviation by a coalition $Z \subseteq N$ that implements a network $h \neq g$. Using the expressions for the profits of suppliers and retailers provided in (5) and (6), the fact that the

deviation is mutually profitable implies that, for all $s \in Z$,

$$\sum_{r \in R} \ell^h_{sr} t^h_{sr} > \sum_{r \in R} \ell^g_{sr} t^g_{sr}, \tag{A-4}$$

and, for all $r \in Z$,

$$\Pi_r^h - \sum_{s \in S} \ell_{sr}^h t_{sr}^h > \Pi_r^g - \sum_{s \in S} \ell_{sr}^g t_{sr}^g.$$
(A-5)

Note that, for $k \in \{h, g\}$, one can write $\sum_{r \in R} \ell_{sr}^k t_{sr}^k = \sum_{r \in Z} \ell_{sr}^k t_{sr}^k + \sum_{r \notin Z} \ell_{sr}^k t_{sr}^k$ and $\sum_{s \in S} \ell_{sr}^k t_{sr}^k = \sum_{s \in Z} \ell_{sr}^k t_{sr}^k + \sum_{s \notin Z} \ell_{sr}^k t_{sr}^k$. Using these identities and adding (A-4) over $s \in Z$ and (A-5) over $r \in Z$, one obtains

$$\sum_{s\in\mathbb{Z}}\left[\sum_{r\in\mathbb{Z}}\ell^h_{sr}t^h_{sr} + \sum_{r\notin\mathbb{Z}}\ell^h_{sr}t^h_{sr}\right] > \sum_{s\in\mathbb{Z}}\left[\sum_{r\in\mathbb{Z}}\ell^g_{sr}t^g_{sr} + \sum_{r\notin\mathbb{Z}}\ell^g_{sr}t^g_{sr}\right],\tag{A-6}$$

$$\sum_{r\in\mathbb{Z}}\Pi_r^h - \sum_{r\in\mathbb{Z}}\left[\sum_{s\in\mathbb{Z}}\ell_{sr}^h t_{sr}^h + \sum_{s\notin\mathbb{Z}}\ell_{sr}^h t_{sr}^h\right] > \sum_{r\in\mathbb{Z}}\Pi_r^g - \sum_{r\in\mathbb{Z}}\left[\sum_{s\in\mathbb{Z}}\ell_{sr}^g t_{sr}^g + \sum_{s\notin\mathbb{Z}}\ell_{sr}^g t_{sr}^g\right].$$
 (A-7)

Adding up (A-6) and (A-7) and rearranging terms, one obtains

$$\sum_{r\in\mathbb{Z}} \left[\Pi_r^h - \Pi_r^g \right] > \sum_{s\in\mathbb{Z}} \sum_{r\notin\mathbb{Z}} \left(\ell_{sr}^g t_{sr}^g - \ell_{sr}^h t_{sr}^h \right) - \sum_{r\in\mathbb{Z}} \sum_{s\notin\mathbb{Z}} \left(\ell_{sr}^g t_{sr}^g - \ell_{sr}^h t_{sr}^h \right)$$
(A-8)

For all links *sr* in which one of the firms belongs to *Z* and the other does not, as is always the case in the right-hand side of (A-8), one has $\ell_{sr}^h t_{sr}^h = \ell_{sr}^h t_{sr}^g$. This is trivially true for $\ell_{sr}^h = 0$. For $\ell_{sr}^h = 1$, instead, the deviating firm must meet the transfer proposal t_{sr}^g of the firm that does not participate in the deviation, since the latter is not given a chance to modify such proposal and if that proposal were not met one would have $\ell_{sr}^h = 0$. This implies that $\ell_{sr}^h = 1$ if and only if $t_{sr}^h = t_{sr}^g$. The fact that $\ell_{sr}^h t_{sr}^h = \ell_{sr}^h t_{sr}^g$ in the right-hand side of (A-8) yields the result in (7).

Next, I show that (7) is a sufficient condition for there to exist a mutually profitable deviation by a coalition $Z \in \mathbb{Z}_{g \to h}$ to a network h with transfers t^h . For each $r \in Z$, choose a profile of links ℓ^h_{sr} and transfers t^h_{sr} with $s \notin Z$ such that (A-5) just holds, so that all retailers in the deviating coalition are just better off in network h than in network g. That is, for each $r \in Z$, choose $\sum_{s \notin Z} \ell^h_{sr} t^h_{sr}$ and an arbitrarily small $\varepsilon_r > 0$ such that

$$\sum_{s \notin Z} \left(\ell_{sr}^g t_{sr}^g - \ell_{sr}^h t_{sr}^h \right) = \left(\Pi_r^g - \Pi_r^h \right) - \sum_{s \in Z} \left(\ell_{sr}^g t_{sr}^g - \ell_{sr}^h t_{sr}^h \right) - \varepsilon_r.$$
(A-9)

If (7) holds then, for the reasoning above, so does (A-8). Substituting (A-9) into (A-8) and simplifying one obtains

$$\sum_{s\in\mathbb{Z}}\left[\sum_{r\in\mathbb{R}}\left(\ell_{sr}^{h}t_{sr}^{h}-\ell_{sr}^{g}t_{sr}^{g}\right)\right]>\sum_{r\in\mathbb{Z}}\varepsilon_{r}.$$
(A-10)

The terms in square brackets in the left-hand side of (A-10) are independent from one another, since each refers to a different supplier $s \in Z$, and can be chosen freely, since the restrictions imposed by (A-9) constrain ℓ_{sr}^h and t_{sr}^h for $s \notin Z$ but not for $s \in Z$. Therefore, for $\sum_{r \in Z} \varepsilon_r \to 0^+$, (A-10) implies that one can always find ℓ_{sr}^h and t_{sr}^h for each $s \in Z$ such that (A-4) holds for all $s \in Z$. This proves that, if (7) holds, one can always find a network *h* with profile of transfers t^h such that all suppliers and retailers in some coalition $Z \in \mathcal{Z}_{g \to h}$ are better off.

Proof of Proposition 2 (No exclusive contracts, with transfers) – The proof proceeds by first characterizing the conditions under which there exist coalition-proof Nash equilibria ("equilibria", for short) with $g_{t,ne}^* = pe$ and $g_{t,ne}^* = dca$, and then proving that there never exist equilibria with $g_{t,ne}^* \in \{bm, dm, um, mix\}$.

Existence of equilibria with $g_{t,ne}^* = pe$

Consider a candidate equilibrium with g = pe (see Figure 1(c)), in which no firm is willing to engage in "cross trade," i.e., in which *s* and *r*' refuse to trade with each other and the same applies to *s*' and *r*. Equilibria with g = pe can also be supported by having only one of *s* and *r*' (or one of *s*' and *r*) refuse to trade with the other. However, such candidate equilibria would be easier to break than candidate equilibria supported by strategies in which both firms refuse to trade with the other. The restriction to the latter is, therefore, without loss of generality. Consider a symmetric profile of transfers $t_{sr} = t_{s'r'} = t^{pe}$ for the pairs of firms that agree to trade. This restriction is also without loss of generality because candidate equilibria with asymmetric transfers are easier to break than candidate equilibria with symmetric transfers, since, when transfers are asymmetric, the supplier and/or retailer with the lowest payoff would be more amenable to a deviation.

For there to be no profitable unilateral deviations in which a supplier or a retailer rescinds his or her link it must be

$$0 \le t^{pe} \le \Pi^{pe}. \tag{A-11}$$

No firm can unilaterally create a new link, since this requires consent from a supplier-retailer pair. Consider then joint deviations by supplier *s* and retailer r'. This pair can implement the following deviations:

A deviation to g = dm, with *s* agreeing to trade with *r'* and stop trading with *r*, while *r'* continues to trade with *s'*. For $t^{pe} > 0$, this deviation is not self-enforcing in the absence of exclusive contracts, since *s* would always have an incentive to re-form his link with *r*. For $t^{pe} = 0$, this deviation is instead self enforcing, since *s* would have no incentive to reform his link with

r, and mutually profitable, since $\Pi^{dm} \ge \Pi^{pe}$. For an equilibrium with g = pe to exist it must therefore be $t^{pe} > 0$.

A deviation to g = mix, with r' agreeing to trade with both s and s', while s continues to trade with r. This deviation is self-enforcing (i.e., supportable as a Nash equilibrium of the two-player game between s and r') if and only if $t^{pe} \leq \Pi^{mix2} - \Pi^{um}$ and strictly Pareto dominates g = pe if and only if $\Pi^{mix2} > \Pi^{pe}$. Therefore, there does not exist a mutually profitable and self-enforcing deviation to g = mix if and only if at least one of the following holds

$$\Pi^{pe} \ge \Pi^{mix2} \tag{A-12}$$

$$t^{pe} > \Pi^{mix2} - \Pi^{um} \tag{A-13}$$

A deviation to g = um, with r' agreeing to start trading with s and stop trading with s', while s continues to trade with r. This deviation is self-enforcing if and only if $t^{pe} \ge \Pi^{mix2} - \Pi^{um}$ and Pareto dominates g = pe if and only if $t^{pe} > \Pi^{pe} - \Pi^{um}$. This implies that there exists no mutually profitable and self-enforcing deviation to g = um if and only if

$$t^{pe} \le \max\left\{\Pi^{mix2}, \Pi^{pe}\right\} - \Pi^{um},\tag{A-14}$$

with strict inequality if $\Pi^{mix2} > \Pi^{pe}$.

One can summarize the analysis above as follows. If (A-12) does not hold, there never exists an equilibrium with g = pe, since both (A-13) and (A-14) would need to hold, and there does not exist any t^{pe} for which this can be the case. If instead (A-12) holds, the only other condition required for there to exist no self-enforcing deviation by $\{s, r'\}$ or $\{s', r\}$ is that (A-14) be satisfied by some $t^{pe} > 0$, which is always the case since $\Pi^{pe} > \Pi^{um}$. Therefore, for $\Pi^{pe} \ge \Pi^{mix2}$, there always exist profiles of transfers $t^{pe} \in (0, \Pi^{pe} - \Pi^{um}]$ that make g = pe immune to deviations by $\{s, r'\}$ or $\{s', r\}$.

Given that, in the candidate equilibrium, *s* and *r'*, as well as *s'* and *r*, refuse to trade with each other, coalitions of three firms can do no better than $\{s, r'\}$ or $\{s', r\}$. The only coalition that can achieve deviations not available to $\{s, r'\}$ or $\{s', r\}$ is the grand coalition $\{s, s', r, r'\}$, which can implement g = dca. Such a deviation is, however, never profitable for all firms in $\{s, s', r, r'\}$ when $\Pi^{pe} \ge \Pi^{mix2}$, since, with linear demand, this condition implies $\Pi^{pe} > \Pi^{dca}$.

Taken together, all of the above implies that there exist equilibria with g = pe and $t^{pe} \in (0, \Pi^{pe} - \Pi^{um}]$ if and only if $\Pi^{pe} \ge \Pi^{mix2}$.

Existence of equilibria with $g_{t,ne}^* = dca$

Consider a candidate equilibrium with g = dca as in Figure 1(a) and $t_{sr}^{dca} = t^{dca}$ for all s and

r. As explained above, the restriction to symmetric transfers is without loss of generality. For a supplier or retailer to have no incentives to rescind unilaterally one of their supply links, thus implementing a market structure with g = mix as in Figure 1(b), it must be

$$0 \le t^{dca} \le \Pi^{dca} - \Pi^{mix1}. \tag{A-15}$$

Moreover, when (A-15) holds, in the candidate equilibrium a retailer earns $\Pi^{dca} - 2t^{dca} \ge 2\Pi^{mix1} - \Pi^{dca} > 0$ (where the last inequality follows from the products being substitutes) and has thus no incentive to withdraw from the market completely by rescinding both of its links (a supplier has obviously no incentives to do so for $t^{dca} \ge 0$.)

Consider a deviation to g = pe by the grand coalition $\{s, s', r, r'\}$ in which all firms refuse to engage in cross trade (e.g., *s* and *r'* refuse to deal with each other and *s'* and *r* refuse to deal with each other). Two aspects of this deviation are worth mentioning. First, as discussed in relation to Remark 1 in the main text, this deviation could be viewed as a (nonbinding) market allocation agreement. The consequences of prohibiting such agreements (and thus the deviation considered here) are discussed in Remark 1, which is proved further below in this appendix. For the time being, however, I allow for this type of nonbinding agreements. Second, a deviation to g = pe can also be implemented by smaller coalitions (e.g., $\{r, r'\}$ agreeing not to "cross trade" with suppliers) but such deviations would be less likely to be self enforcing than a deviation by $\{s, s', r, r'\}$, as in such deviations not all firms would withdraw their cross offers (e.g., in a deviation by $\{r, r'\}$ only the retailers would withdraw their cross offers, whereas the suppliers would continue to offer contracts to all retailers). Such deviations by smaller coalitions would thus not add anything to the analysis presented here. Having clarified these aspects, denote by \tilde{t}^{pe} the transfers in this deviation. The deviation can be mutually profitable if there exists a \tilde{t}^{pe} such that

$$2t^{dca} < \tilde{t}^{pe} < \Pi^{pe} - \Pi^{dca} + 2t^{dca} \tag{A-16}$$

The deviation is never self enforcing for $\Pi^{mix^2} > \Pi^{pe}$ because, as shown above, when this condition holds a network with g = pe is never self-enforcing even with unrestricted transfers t^{pe} . Therefore, it is a fortiori never self-enforcing with transfers \tilde{t}^{pe} that are restricted as in (A-16). The deviation is, instead, always self enforcing for $\Pi^{pe} \ge \Pi^{mix^2}$, as there always exists a $\tilde{t}^{pe} > 0$ that satisfies both (A-14) and (A-16). The latter requires $\Pi^{pe} > \Pi^{dca}$ (which is always the case when $\Pi^{pe} \ge \Pi^{mix^2}$, as $\Pi^{mix^2} > \Pi^{dca}$) and $2t^{dca} < \Pi^{pe} - \Pi^{um}$. For there to be no $\tilde{t}^{pe} > 0$ that satisfies (A-14) and (A-16) one would therefore need $2t^{dca} \ge \Pi^{pe} - \Pi^{um}$. However, with linear demand and when $\Pi^{pe} \ge \Pi^{mix^2}$, there never exists a t^{dca} satisfying both this condition and condition (A-15) for no unilateral deviations.

Consider next a deviation to g = dm by a coalition $\{s, s', r\}$. For $t^{dca} > 0$, this deviation is not self-enforcing in the absence of exclusive contracts, since either supplier would always have incentives to re-form his link with r'. For $t^{dca} = 0$, this deviation is instead self enforcing, since no supplier would have incentives to reform his link with r', and mutually profitable, since $\Pi^{dm} \ge \Pi^{dca}$. For an equilibrium with g = dca to exist it must therefore be $t^{dca} > 0$.

Finally, consider a deviation to g = um by a coalition $\{s, r, r'\}$. This deviation is self enforcing if and only if $t^{dca} > \Pi^{mix2} - \Pi^{um}$ and always mutually profitable whenever it is self enforcing. A necessary condition for there to exist an equilibrium with g = dca is therefore $t^{dca} \leq \Pi^{mix2} - \Pi^{um}$. Since with linear demand $\Pi^{mix2} \geq \Pi^{dca}$ and $\Pi^{um} = \Pi^{mix1}$, this condition is never binding when (A-15) holds.

Taken together, all of the above implies that there exist equilibria with g = dca and $t^{dca} \in (0, \Pi^{dca} - \Pi^{mix1}]$ if and only if $\Pi^{mix2} > \Pi^{pe}$.

There do not exist equilibria with $g_{t,ne}^* \in \{bm, dm, um, mix\}$

There cannot exist an equilibrium with g = bm because the excluded supplier and retailer would always have self-enforcing incentives to form a link and earn positive profits.

There cannot exist an equilibrium with g = dm because the excluded retailer, r', can always profitably deviate by entering into a contract with at least one supplier. Without exclusive contracts, that supplier would be able to keep his contract with the other retailer, r, and would thus be willing to accept any additional positive transfer from r'. Analogously, there cannot exist an equilibrium with g = um either. The excluded supplier, s', can always profitably deviate by offering a contract with a positive and arbitrarily small transfer request to at least one of the retailers. Since that retailer can keep his link with s and since $\Pi^{mix2} > \Pi^{um}$, he accepts the proposal by s'.

Finally, consider a candidate mixed equilibrium with g = mix where all links are active except the link between s' and r'. There always exists a self-enforcing deviation in which s' and r' form that link. If in the candidate equilibrium $t_{sr'}^{mix} \leq \Pi^{dca} - \Pi^{mix1}$, the self-enforcing deviation involves r' maintaining her link with s and implementing g = dca (this deviation is jointly profitable for s' and r' since $\Pi^{dca} > \Pi^{mix1}$). If instead in the candidate equilibrium $t_{sr'}^{mix} > \Pi^{dca} - \Pi^{mix1}$, the self-enforcing deviation involves r' rescinding her link with s and implementing g = mix (this deviation, which effectively involves switching s' for s, is jointly profitable for s' and r' since $t_{sr'}^{mix} > \Pi^{dca} - \Pi^{mix1} > 0$). As a result, there never exists an equilibrium with a mixed network.

Proof of Remark 1 (No exclusive contracts, no market allocation agreements) – Regarding the existence of equilibria with g = pe, as the conditions obtained in the proof of Proposition 2 do

not rely on market allocation agreements, prohibiting such agreements has no effects.

Regarding the existence of equilibria with g = dca, prohibiting market allocation agreements rules out the deviation to g = pe by the grand coalition in the proof of Proposition 2. The only feasible deviation to g = pe remains one by a supplier *s* and a retailer *r*, but such a deviation is not self enforcing as, without exclusive contracts, the supplier has always incentives to reform his link with the other retailer, r'. Since, as established in the proof of Proposition 2, there always exists a t^{dca} that prevents self-enforcing deviations to networks other than g = pe, there always exist equilibria with g = dca.

Proof of Proposition 3 (Exclusive contracts and transfers) – The proof proceeds by first characterizing the conditions under which there exist equilibria with $g_{t,e}^* \in \{dca, pe, dm\}$ and then proving that there never exist equilibria with $g_{t,e}^* \in \{bm, um, mix\}$.

Existence of equilibria with $g_{t,e}^* = dca$

As in the proof of Proposition 2, one can focus without loss of generality on symmetric equilibria with $t_{sr}^{dca} = t^{dca}$ for all *s* and *r*. Suppliers and retailers do not find it profitable to rescind unilaterally one of their links, thus implementing g = mix, if and only if (A-15) holds. Deviations to g = mix can be also carried out by coalitions including more than one firm, but such deviations would be less likely to be jointly profitable or self-enforcing than a unilateral deviation and thus do not play any role when (A-15) holds.

Consider a deviation by a coalition $\{s, r\}$ in which these two firms commit to mutual exclusivity, thus implementing g = pe. This deviation is mutually profitable if and only if there exists a \tilde{t}^{pe} such that

$$2t^{dca} < \tilde{t}^{pe} < \Pi^{pe} - \Pi^{dca} + 2t^{dca}$$
(A-17)

When mutually profitable, this deviation is always self-enforcing. To see why, note that the adoption of mutual exclusivity by $\{s, r\}$ implies that if either *s* or *r* deviates further it must stop trading with the other. The only possible further deviations from g = pe are, therefore, *s* trading only with *r'* (thus implementing g = dm, since *s'* continues to trade with *r'*) or *r* trading only with *s'* (thus implementing g = um, since *r'* continues to trade with *s'*.) These deviations are not mutually profitable, and thus the original deviation is self-enforcing, if and only if

$$t^{dca} \le \tilde{t}^{pe} \le \Pi^{pe} - \Pi^{um} + t^{dca} \tag{A-18}$$

Since in any equilibrium with g = dca condition (A-15) must hold and since with linear demand $\Pi^{um} = \Pi^{mix1}$, it is straightforward to verify that (A-18) always holds when (A-17) holds, and thus the original deviation to g = pe is always self-enforcing when mutually profitable. Using (A-17), this implies that there does not exist a self-enforcing deviation to g = pe by $\{s, r\}$ if and only if

$$\Pi^{dca} \ge \Pi^{pe} \tag{A-19}$$

Deviations to g = pe can also be implemented by broader coalitions, such as the grand coalition $\{s, s', r, r'\}$ in the proof of Proposition 2, but these deviations do not add anything to the conditions derived from the deviation by $\{s, r\}$ studied above.

Consider now a deviation to g = um in which r and r' commit to be exclusive to s. This deviation is mutually profitable if and only if the joint profits that $\{s, r, r'\}$ can earn in the deviation, $2\Pi^{um}$, are greater than the joint profits that $\{s, r, r'\}$ earns in the candidate equilibrium, $2\Pi^{dca} - 2t^{dca}$, which is the case if and only if $t^{dca} > \Pi^{dca} - \Pi^{um}$. This condition never holds, and thus a deviation to g = um is never profitable, when condition (A-15) for no unilateral deviations holds, since, with linear demand, $\Pi^{um} = \Pi^{mix1}$.

Finally, consider a deviation to g = dm in which *s* and *s'* commit to be exclusive to *r*. This deviation is mutually profitable if and only if there exists a \tilde{t}^{dm} such that

$$4t^{dca} < 2\tilde{t}^{dm} < \Pi^{dm} - \Pi^{dca} + 2t^{dca}.$$
(A-20)

Moreover, the deviation is self enforcing if and only if none of the following subsequent deviations is profitable and self enforcing: i) one of the suppliers rejects r's proposal and accepts the candidate equilibrium proposal t^{dca} of r', which implements g = pe; ii) a coalition of one supplier and one retailer, say $\{s, r\}$, agrees to mutual exclusivity, with r rejecting the deviation proposal of s', which, after s' enters a new contract with r', implements g = pe; iii) a coalition of one supplier and one retailer, say $\{s, r\}$, agrees that r rejects the deviation proposal of s', while s accepts the candidate equilibrium proposal t^{dca} of r', which, after s' enters a new contract with r' as a consequence of the "no stranding" assumption, implements g = mix; iv) a coalition of one supplier and one retailer, say $\{s, r\}$, agrees that *r* accepts the deviation proposal of s', while s accepts the candidate equilibrium proposal t^{dca} of r', which implements g = mix. Subsequent deviations i) and ii) are not profitable, and thus the original deviation to g = dm is self enforcing, if and only if $2t^{dca} < 2\tilde{t}^{dm} < 2(\Pi^{dm} - \Pi^{pe})$. In the relevant case where $t^{dca} \ge 0$ (see condition (A-15) above) and $\Pi^{dca} \ge \Pi^{pe}$ (see condition (A-19) above) this is always true when \tilde{t}^{dm} satisfies (A-20), i.e., when the initial deviation to g = dm is mutually profitable. Deviations iii) and iv) are instead never mutually profitable and self enforcing when the original deviation to g = dmis mutually profitable and condition (A-15) for the existence of an equilibrium with $g_{t,e}^* = dca$ holds. These deviations do not, therefore, add anything to the analysis. The detailed proof of this fact is tedious and is contained in the online appendix. All of this implies that, for there to

exist no mutually profitable and self-enforcing deviation to g = dm, (A-20) must fail for any \tilde{t}^{dm} , which is the case if and only if

$$2t^{dca} \ge \Pi^{dm} - \Pi^{dca} \tag{A-21}$$

The analysis above implies that there exist equilibria with g = dca if and only if (A-19) holds and there exists a t^{dca} such that (A-15) and (A-21) also hold. The latter is the case if and only if

$$2\left(\Pi^{dca} - \Pi^{mix1}\right) \ge \Pi^{dm} - \Pi^{dca}.$$
(A-22)

Intuitively, for there to exist an equilibrium with g = dca there must exist intermediate transfers t^{dca} that ensure that i) either supplier does not extract too much, otherwise one or both retailers would drop him and ii) either retailer does not extract too much, otherwise the other retailer could convince both suppliers to drop her. As can be seen in Figure 4, this is possible only if suppliers and retailers are sufficiently differentiated.

Existence of equilibria with $g_{t,e}^* = pe$

Consider an equilibrium with g = pe supported by mutual exclusivity between s and r and between s' and r'. Although equilibria with g = pe can also be supported by one-way commitments to exclusivity, equilibria supported by mutual exclusivity are more difficult to break and thus more likely to exist. As in the proof of Proposition 2, focus without loss of generality on symmetric transfers $t_{sr} = t_{s'r'} = t^{pe}$ for the pairs of firms that agree to trade. Suppliers and retailers do not have incentives to deviate unilaterally by becoming inactive if and only if (A-11) holds.

Consider first a deviation to g = um in which r and r' commit to be exclusive to s. This deviation is mutually profitable if and only if there exists a \tilde{t}^{um} such that $t^{pe} < 2\tilde{t}^{um} < 2(\Pi^{um} - \Pi^{pe} + t^{pe})$ and is (trivially) self enforcing whenever it is profitable (in the candidate equilibrium, supplier s' does not have a proposal out to retailer r and cannot be involved in further negotiations because it does not belong to the original deviating coalition that includes r). Therefore, for there to exist no mutually profitable and self-enforcing deviation to g = um it must be

$$t^{pe} \le 2\left(\Pi^{pe} - \Pi^{um}\right) \tag{A-23}$$

Consider next a deviation to g = dm in which s and s' commit to be exclusive to r and r pays a transfer \tilde{t}^{dm} to each supplier. This deviation is mutually profitable if and only if there exists a \tilde{t}^{dm} such that $2t^{pe} < 2\tilde{t}^{dm} < \Pi^{dm} - \Pi^{pe} + t^{pe}$ and, as above, is (trivially) self enforcing whenever it is profitable (in the candidate equilibrium, retailer r' does not have a proposal out to supplier s and

cannot be involved in further negotiations because it does not belong to the original deviating coalition that includes *s*). Therefore, for there to exist no mutually profitable and self-enforcing deviation to g = dm it must be

$$t^{pe} \ge \Pi^{dm} - \Pi^{pe}. \tag{A-24}$$

Finally, consider deviations to g = mix or g = dca. Since the candidate equilibrium with g = pe is supported by mutual exclusivity, these deviations can only be implemented by the grand coalition $\{s, s', r, r'\}$, because all the contract proposals in the candidate equilibrium would need to be modified. This deviation can therefore be mutually profitable only if, respectively, $\Pi^{mix1} + \Pi^{mix2} > 2\Pi^{pe}$ and $2\Pi^{dca} > 2\Pi^{pe}$. With linear demand, this is never the case in the region of parameters where (A-23) and (A-24) hold.

In conclusion, there exists an equilibrium with g = pe if and only if there exists a t^{pe} such that (A-11), (A-24) and (A-23) hold, which is the case if and only if

$$2\Pi^{pe} \ge \Pi^{dm},\tag{A-25}$$

$$2\left(\Pi^{pe} - \Pi^{um}\right) \ge \Pi^{dm} - \Pi^{pe}.\tag{A-26}$$

Condition (A-25) follows from the need for t^{pe} to satisfy (A-11) and (A-24), which can be the case only if pairwise exclusivity yields higher profits than downstream monopoly, whereas condition (A-26) follows from the need for t^{pe} to satisfy (A-23) and (A-24) (the requirement that $t^{pe} \ge 0$ is satisfied when (A-24) holds because $\Pi^{dm} - \Pi^{pe} > 0$).

Existence of equilibria with $g_{t,e}^* = dm$

Consider a candidate equilibrium with g = dm in which s and s' commit to be exclusive to r and assume, without loss of generality, symmetric transfers $t_{sr}^{dm} = t_{s'r}^{dm} = t^{dm}$. For there to be no jointly profitable and self-enforcing deviation in which the excluded retailer, r', and one of the suppliers enter a mutually exclusive contract that implements g = pe it must be $t^{dm} \ge \Pi^{pe}$. Moreover, for there to be no jointly profitable and self-enforcing deviation in which the monopolistic retailer, r, and one of the two suppliers, say s, enter a mutually exclusive contract it must be $t^{dm} \le \Pi^{dm} - \Pi^{pe}$ (note that, by the "no stranding" assumption, the rejection of s''s exclusive proposal by r would allow s' to enter a new contract with r' and the market structure resulting from this deviation would thus be g = pe). There exists a t^{dm} that satisfies these two conditions if and only if (A-25) fails (or holds with equality).

In this equilibrium both retailers must earn zero profits, which implies that the active retailer must pay transfers $t^{dm} = \Pi^{dm}/2$. If $t^{dm} < \Pi^{dm}/2$, there would exist a profitable deviation

in which the excluded retailer, r', steals the entire business of the active retailer, r, by offering both suppliers transfers $\tilde{t}^{dm} \in (t^{dm}, \Pi^{dm}/2]$ in exchange for exclusivity and earns profits $\Pi^{dm} - 2\tilde{t}^{dm} > 0$. Such a deviation would be self enforcing because the only feasible further deviation, i.e., r' and one of the two suppliers, say s', committing to mutual exclusivity, is not profitable. This is the case because, by the "no stranding" assumption, such a deviation would cause the two excluded firms, s and r, to enter a contract and would thus implement g = pe, and $\Pi^{pe} \leq \Pi^{dm} - \tilde{t}^{dm}$ for $\tilde{t}^{dm} \leq \Pi^{dm}/2$ when $\Pi^{dm} \geq 2\Pi^{pe}$.

Finally, when (A-25) fails (or holds with equality), one also has that $\Pi^{dm} > 2\Pi^{dca}$ and $\Pi^{dm} > \Pi^{mix1} + \Pi^{mix2}$ so that deviations in which the grand coalition deviates to g = dca or a coalition of $\{s', r, r'\}$ deviates to g = mix (with r' carrying product s' and s carrying both products) are not mutually profitable.

There do not exist equilibria with $g_{t,e}^* \in \{bm, um, mix\}$

There cannot exist an equilibrium with g = bm because the excluded supplier and retailer would always have self-enforcing incentives to form a link and earn positive profits.

Consider a candidate equilibrium with g = um in which r and r' commit to be exclusive to s and assume, without loss of generality, symmetric transfers $t_{sr}^{um} = t_{sr'}^{um} = t^{um}$. Unless $t^{um} \leq \Pi^{um} - \Pi^{pe}$ there always exists a self-enforcing deviation in which the excluded supplier, s', and one of the retailers, say r, enter a mutually exclusive contract that implements g = pe. Since $\Pi^{um} - \Pi^{pe} < 0$ and suppliers have incentives to rescind their candidate equilibrium supply link if $t^{um} < 0$, there does not exist any equilibrium with g = um.

Finally, consider a candidate equilibrium with g = mix in which all links are active except the link between s' and r'. Specifically, s' commits to exclusivity with r, r' commits to exclusivity with s, and s and r trade with each other on a nonexclusive basis. In such a candidate equilibrium it must be

$$t_{st'}^{mix} = 0. \tag{A-27}$$

To prove this, assume by contradiction that in equilibrium $t_{sr'}^{mix} > 0$. A coalition $\{s', r, r'\}$ can then engage in a mutually profitable deviation to a different mixed network with s' switched for s and transfers $\tilde{t}_{sr}^{mix} = t_{sr}^{mix}$, $\tilde{t}_{s'r}^{mix} = t_{s'r}^{mix}$ and $\tilde{t}_{s'r'}^{mix} = t_{sr'}^{mix} - \varepsilon$, $\varepsilon > 0$. Given symmetry of demand, the network resulting from this deviation yields the same structure of payoffs (with s' switched for s) as the equilibrium network, up to an arbitrarily small difference ε . Since the equilibrium network is self-enforcing by definition, this deviation is also self-enforcing, which contradicts the assumption that $t_{sr'}^{mix} > 0$ constitutes an equilibrium. It must therefore be $t_{sr'}^{mix} \leq 0$. Moreover, for supplier s not to rescind his link with r' unilaterally it must be $t_{sr'}^{mix} \geq 0$. This establishes (A-27). Intuitively, as the suppliers are symmetric, either supplier can provide the same profits by supplying his product to the retailer that carries only one product, and no supplier can thus extract any profits by doing so.

Consider now a deviation by a coalition $\{s, r\}$ in which *s* commits to be exclusive to *r*, thus implementing g = dm. This deviation is jointly profitable if and only if there exists a \tilde{t}_{sr}^{dm} such that

$$t_{sr'}^{mix} + t_{sr}^{mix} < \tilde{t}_{sr}^{dm} < \Pi^{dm} - \Pi^{mix2} + t_{sr}^{mix},$$
(A-28)

and self-enforcing if and only if neither s nor r find it profitable to deviate further by stopping dealing with the other, which is the case if and only if

$$t_{sr'}^{mix} \le \tilde{t}_{sr}^{dm} \le \Pi^{dm} - \Pi^{pe}.$$
(A-29)

Since in any candidate equilibrium with g = mix it must be $t_{sr}^{mix} \ge 0$, (A-28) and (A-29) can be consolidated into the following condition for the existence of a profitable and self-enforcing deviation to g = dm

$$t_{sr'}^{mix} + t_{sr}^{mix} < \min\left\{\Pi^{dm} - \Pi^{mix2} + t_{sr}^{mix}, \Pi^{dm} - \Pi^{pe}\right\},$$
(A-30)

If t_{sr}^{mix} is such that the right-hand side of (A-30) is $\Pi^{dm} - \Pi^{mix2} + t_{sr}^{mix}$, for there to exist no profitable and self-enforcing deviation to g = dm it must be $t_{sr'}^{mix} \ge \Pi^{dm} - \Pi^{mix2}$, which can never be the case, since $\Pi^{dm} - \Pi^{mix2} > 0$ and (A-27) must hold. If instead t_{sr}^{mix} is such that the right-hand side of (A-30) is $\Pi^{dm} - \Pi^{pe}$, for there to exist no profitable and self-enforcing deviation to g = dm it must be $t_{sr'}^{mix} \ge \Pi^{dm} - \Pi^{pe} - t_{sr}^{mix}$. Since in any equilibrium with g = mix it must be $t_{sr'}^{mix} \le \Pi^{dm} - \Pi^{pe}$ (otherwise r would drop s), the previous condition implies, again, $t_{sr'}^{mix} \ge \Pi^{dm} - \Pi^{mix2}$, which, as above, can never be the case. Therefore, when (A-27) holds there always exist profitable and self-enforcing deviations to g = dm by $\{s, r\}$. Intuitively, since s is not compensated for providing his product to the single-product retailer, r', he can be easily induced to help the two-product retailer, r, monopolize the downstream market by not selling to r'.

Proof of Proposition 5 (Distribution of profits) – It is straightforward to see that the availability of exclusive contracts can never make suppliers unambiguously worse off, as the lower bound on suppliers' profits without exclusive contracts is always zero (see Proposition 2) and suppliers cannot be forced to earn negative profits. The fact that the availability of exclusive contracts can (and, for *b* sufficiently high, does) make suppliers unambiguously better off and retailers unambiguously worse off is, however, less immediate.

Consider first the case in which exclusive contracts result in an equilibrium supply network

with $g_{t,e}^* = pe$. There are two cases: i) the equilibrium supply network is $g_{t,e}^* = g_{t,ne}^* = pe$ regardless of whether exclusive contracts are available or not, which is the case for $\overline{b}_{t,ne}(a) \leq b \leq \overline{b}_m(a)$ (note that this case makes sense only with Bertrand downstream competition, as with Cournot downstream competition there is no *b* for which $g_{t,ne}^* = pe$ is an equilibrium without exclusive contracts); ii) exclusive contracts cause the equilibrium supply network to switch from $g_{t,ne}^* = dca$ to $g_{t,e}^* = pe$, which is the case for $\overline{b}_{t,e}(a) \leq b < \overline{b}_{t,ne}(a)$.

Case i): As total industry profits remain the same, an unambiguous increase in suppliers' profits implies an unambiguous decrease in retailers' profits (and vice versa). It is therefore sufficient to analyze the profits of only one type of firm. In the absence of exclusive contracts, suppliers' profits have an upper bound $\overline{\pi}_{s,ne}^{pe} = \Pi^{pe} - \Pi^{um}$, whereas when exclusive contracts are available they have a lower bound $\underline{\pi}_{s,e}^{pe} = \Pi^{dm} - \Pi^{pe}$. With linear demand $\underline{\pi}_{s,e}^{pe} > \overline{\pi}_{s,ne}^{pe}$, and suppliers are therefore unambiguously better off (with retailers being unambiguously worse off) when exclusive contracts are available.

Case ii): As the equilibrium changes, so do total industry profits. One needs therefore to carry out separate profit comparisons for suppliers and retailers. Suppliers' profits have an upper bound $\overline{\pi}_{s,ne}^{dca} = 2\left(\Pi^{dca} - \Pi^{mix1}\right)$ without exclusive contracts and a lower bound $\underline{\pi}_{s,e}^{pe} = \Pi^{dm} - \Pi^{pe}$ with exclusive contracts. Retailers' profits have a lower bound $\underline{\pi}_{r,ne}^{dca} = 2\Pi^{mix1} - \Pi^{dca}$ without exclusive contracts and an upper bound $\overline{\pi}_{r,e}^{pe} = 2\Pi^{pe} - \Pi^{dm}$ with exclusive contracts. Retailers' profits have a lower bound $\underline{\pi}_{r,ne}^{dca} = 2\Pi^{mix1} - \Pi^{dca}$ without exclusive contracts and an upper bound $\overline{\pi}_{r,e}^{pe} = 2\Pi^{pe} - \Pi^{dm}$ with exclusive contracts. With linear demand, $\underline{\pi}_{s,e}^{pe} > \overline{\pi}_{s,ne}^{dca}$ in the region of parameters where $g_{t,e}^* = pe$ (see point 2 in Proposition 3) and $\overline{\pi}_{r,e}^{pe} < \underline{\pi}_{r,ne}^{dca}$ for all parameter values, which implies that the availability of exclusive contracts makes suppliers unambiguously better off and retailers unambiguously worse off.

Consider next the case in which exclusive contracts cause the equilibrium supply network to switch to $g_{t,e}^* = dm$, which is the case for $b \ge \overline{\overline{b}}_m(a)$. The analysis differs between i) Bertrand competition, for which the equilibrium network switches from $g_{t,ne}^* = pe$ to $g_{t,e}^* = dm$; ii) Cournot competition, for which the equilibrium network switches from $g_{t,ne}^* = dca$ to $g_{t,e}^* = dm$.

Case i) (Bertrand competition): Suppliers' profits have an upper bound $\overline{\pi}_{s,ne}^{pe} = \Pi^{pe} - \Pi^{um}$ without exclusive contracts and a lower bound $\underline{\pi}_{s,e}^{dm} = \Pi^{dm}/2$ with exclusive contracts. In the region where $g_{t,e}^* = dm$ can be supported as an equilibrium, $\Pi^{dm} \ge 2\Pi^{pe}$ and thus $\underline{\pi}_{s,e}^{dm} > \overline{\pi}_{s,ne}^{pe}$, which implies that exclusive contracts make suppliers unambiguously better off. Retailers' profits have a lower bound $\underline{\pi}_{r,ne}^{pe} = \Pi^{um}$ without exclusive contracts and are always equal to zero in an equilibrium with exclusive contracts and $g_{t,e}^* = dm$, which implies that exclusive contracts make retailers unambiguously worse off.

Case ii) (Cournot competition): Suppliers' profits have an upper bound $\overline{\pi}_{s,ne}^{dca}$ =

 $2\left(\Pi^{dca} - \Pi^{mix1}\right)$ without exclusive contracts and a lower bound $\underline{\pi}_{s,e}^{dm} = \Pi^{dm}/2$ with exclusive contracts. With linear demand, in the region where $g_{t,e}^* = dm$ can be supported as an equilibrium, $\underline{\pi}_{s,e}^{dm} > \overline{\pi}_{s,ne}^{dca}$, which implies that exclusive contracts make suppliers unambiguously better off. Retailers' profits have a lower bound $\underline{\pi}_{r,ne}^{dca} = 2\Pi^{mix1} - \Pi^{dca} > 0$ without exclusive contracts and are always equal to zero in an equilibrium with exclusive contracts and $g_{t,e}^* = dm$, which implies that exclusive contracts unambiguously worse off.

Finally, consider the case in which the equilibrium supply network is $g_{t,e}^* = g_{t,ne}^* = dca$ regardless of whether exclusive contracts are available or not, which is the case for $b < \overline{b}_{t,e}(a)$. This is the case when condition (12) in Proposition 3 holds. The lower bound on the suppliers' profits is strictly greater when exclusive contracts are available than when they are not, i.e., $\underline{\pi}_{s,e}^{dca} = \Pi^{dm} - \Pi^{dca} > 0 = \underline{\pi}_{s,ne}^{dca}$, whereas the upper bound is the same regardless of whether exclusive contracts are available or not, i.e., $\overline{\pi}_{s,ne}^{dca} = \overline{\pi}_{s,e}^{dca} = 2 \left(\Pi^{dca} - \Pi^{mix1} \right)$. Therefore, when the equilibrium remains $g_{t,e}^* = g_{t,ne}^* = dca$, the availability of exclusive contracts never makes suppliers worse off and can make them strictly better off. As industry profits are unaffected by the availability of exclusive contracts, the opposite conclusion applies to retailers' profits, which are no higher, and possibly strictly lower, when exclusive contracts are available.

Proof of Proposition 6 (No exclusive contracts, no transfers) – The proof proceeds by first characterizing the conditions under which there exist equilibria with $g_{nt,ne}^* = dca$ and $g_{nt,ne}^* = pe$; and then proving that there never exist equilibria with $g_{nt,ne}^* \in \{bm, dm, um, mix\}$. The expressions for the supplier and retailer profits in network g, π_s^g and π_r^g , are given by (14) in the main text.

Existence of equilibria with $g_{nt,ne}^* = dca$

Since in the candidate equilibrium all firms are willing to trade with all other firms, all deviations involve rescinding links and can be implemented unilaterally by a single supplier or retailer, or by coalitions of two firms, namely $\{s, s'\}$, $\{r, r'\}$ or $\{s, r\}$.

Unilateral deviations are not profitable if and only if $\pi_r^{dca} > \pi_r^{mix1}$ and $\pi_s^{dca} > \pi_s^{mix1}$. The first condition corresponds to

$$\left(2 - \frac{1}{\beta}\right) \left(\Pi^{dca} - \Pi^{mix1}\right) < \Pi^{mix1} \tag{A-31}$$

and always holds for all $\beta \in [0,1]$, since $\Pi^{mix1} - \Pi^{dca} > \Pi^{mix1} > 0$. The second condition corresponds to $2(\Pi^{dca} - \Pi^{mix1}) > \Pi^{mix2} - \Pi^{um}$ and, since with linear demand $\Pi^{mix1} = \Pi^{um}$, can be written as

$$2\Pi^{dca} > \Pi^{mix1} + \Pi^{mix2}. \tag{A-32}$$

Consider then a deviation by the coalition $\{s, s'\}$. This coalition can carry out deviations that implement g = dm, g = mix or g = pe. A deviation that implements g = dm is never self enforcing, since $\pi_s^{pe} > \pi_s^{dm}$ and thus, when supplier *s'* trades only with *r*, supplier *s* prefers trading only with *r'* to also trading only with *r*. A deviation that implements g = mix is equivalent to a unilateral deviation in which one of the suppliers rescinds one of his links and is thus not self enforcing when (A-32) holds. A deviation that implements g = pe is self enforcing if it is a best response for *s* to deal only with *r* when *s'* deals only with *r'*, i.e., if $\pi_s^{pe} > \max\{\pi_s^{mix2}, \pi_s^{dm}\}$. When (A-32) holds, $\pi_s^{mix2} > \pi_s^{dm}$ and this is therefore the case if and only if $2\Pi^{pe} > \Pi^{mix1} + \Pi^{mix2}$. When this deviation is self-enforcing it is also mutually profitable for $\{s, s'\}$, since $\Pi^{pe} > 2(\Pi^{dca} - \Pi^{mix1})$. Therefore, there exists no mutually profitable and self-enforcing deviation by $\{s, s'\}$ if and only if

$$\Pi^{mix1} + \Pi^{mix2} \ge 2\Pi^{pe}. \tag{A-33}$$

In what follows I restrict attention to the region of parameters where (A-33) holds and show that, in this region, there exist no self-enforcing deviations by the coalitions $\{r, r'\}$ and $\{s, r\}$. The coalition $\{r, r'\}$ can carry out deviations that implement g = um, g = mix and g = pe. A deviation that implements g = um is never self-enforcing because $\Pi^{pe} > \Pi^{um}$ implies $\pi_r^{pe} > \pi_r^{um}$; and a deviation to g = mix is never self-enforcing when unilateral deviations are not profitable, i.e. when (A-32) holds. A deviation that implements g = pe is never self-enforcing when (A-33) holds, since (A-33) implies $\pi_s^{mix2} > \pi_s^{pe}$.

A coalition $\{s, r\}$ can carry out a deviation that implements g = pe. As above, such a deviation is never self-enforcing when (A-33) holds, since (A-33) implies $\pi_s^{mix2} > \pi_s^{pe}$. The same coalition can also carry out deviations that implement g = mix but, as argued above, these deviations are never self enforcing when unilateral deviations are unprofitable.

In summary, since (A-32) is not binding when (A-33) holds, (A-33) is the only necessary and sufficient condition for the existence of an equilibrium with $g_{nt,ne}^* = dca$.

Existence of equilibria with $g_{nt,ne}^* = pe$

Consider a candidate equilibrium with g = pe in which *s* is willing to trade only with *r* (and vice versa) and *s'* is willing to trade only with *r'* (and vice versa). As in the proof of Proposition 2, the only relevant deviation is one by the two-firm coalition formed by *s* and *r'* (or *s'* and *r*). This coalition can implement deviations to g = dm, g = um and g = mix. A deviation to g = dm in which *r'* is the only active retailer is not self-enforcing because, when *r'* is willing to deal with both suppliers, *s* prefers exclusivity with *r* to exclusivity with *r'*, since $\Pi^{pe} > \Pi^{dm} - \Pi^{bm}$ implies $\pi_s^{pe} > \pi_s^{dm}$. A deviation to g = um in which *s* is the only active supplier is not self-

enforcing because, when *s* is willing to deal with both suppliers, *r'* prefers exclusivity with *s'* to exclusivity with *s*, since $\Pi^{pe} > \Pi^{um}$ implies $\pi_{r'}^{pe} > \pi_{r'}^{um}$. Finally, a deviation to g = mix is self enforcing if and only if *r'* responds to the willingness of *s* to deal with both retailers by being willing to deal with both suppliers, and vice versa, i.e., if and only if $\pi_s^{mix2} \ge \max\left\{\pi_s^{pe}, \pi_s^{dm}\right\}$ and $\pi_{r'}^{mix2} \ge \max\left\{\pi_{r'}^{pe}, \pi_{r'}^{um}\right\}$. The first conditions corresponds to $\pi_s^{mix2} \ge \pi_s^{pe}$, since $\pi_s^{pe} \ge \pi_s^{dm}$, and is satisfied if and only if (A-33) holds. The second condition corresponds to $\pi_{r'}^{mix2} \ge \pi_{r'}^{um}$, and, rearranging terms, yields

$$\left(2 - \frac{1}{\beta}\right) \left(\Pi^{mix2} - \Pi^{pe}\right) \le \Pi^{um} \tag{A-34}$$

Condition (A-34) always holds when (A-33) does. To see this note that, when (A-33) holds, $\Pi^{mix2} - \Pi^{pe} > 0$, given that $\Pi^{mix1} < \Pi^{pe}$. Since $(2 - 1/\beta)$ is increasing in β , this implies that (A-34) always holds if it holds for $\beta = 1$. The claim then follows from the fact that with linear demand $(\Pi^{mix2} - \Pi^{pe}) \leq \Pi^{um}$ with both Bertrand and Cournot competition. Condition (A-33) is, therefore, the only relevant necessary and sufficient condition for there to exist a self-enforcing deviation to g = mix by the coalition $\{s, r'\}$. Since this deviation is mutually profitable when selfenforcing, there exists an equilibrium with $g^*_{nt,ne} = pe$ if and only if (A-33) fails.

There do not exist equilibria with $g_{nt,ne}^* \in \{bm, dm, um, mix\}$

There cannot exist an equilibrium with g = bm because the excluded supplier and retailer would always have self-enforcing incentives to form a link and earn positive profits. There cannot exist an equilibrium with g = dm in which, say, r is the only active retailer because min $\{\pi_s^{pe}, \pi_s^{mix2}\} > \pi_s^{dm}$ and thus one of the suppliers can profit by opening a link with r', who would also profit from the deviation. There cannot exist an equilibrium with g = um in which, say, s is the only active supplier because min $\{\pi_r^{pe}, \pi_r^{mix2}\} > \pi_r^{um}$ and thus one of the retailers can profit by opening a link with s', who would also profit from the deviation.

Finally, consider a candidate equilibrium with g = mix in which the link between, say, s and r is inactive. If (A-32) holds there exists a mutually profitable and self-enforcing deviation in which s and r form a link with each other, thus implementing g = dca. If (A-33) holds there exists a mutually profitable and self-enforcing deviation in which s' and r' rescind their link, thus implementing g = pe. Since one of (A-32) and (A-33) always holds, there does not exist any equilibrium with g = mix.

Proof of Proposition 7 (Exclusive contracts, no transfers) – The proof proceeds by first characterizing the conditions under which there exist equilibria with $g_{nt,e}^* = dca$, $g_{nt,e}^* = pe$ and $g_{nt,e}^* = mix$; and then proving that there never exist equilibria with $g_{nt,e}^* \in \{bm, dm, um\}$. *Existence of equilibria with* $g_{nt,e}^* = dca$

First, note that all the deviations that were profitable and self-enforcing without exclusive contracts remain so with exclusive contracts. Therefore, in light of Proposition 6, an equilibrium with g = dca can exist only for $b \leq \overline{b}_{nt,ne}(a)$. Exclusive contracts may also render additional deviations self-enforcing.

A deviation by a coalition $\{s, r\}$ in which these two firms commit to mutual exclusivity, thus implementing g = pe, is profitable for *s* if and only if

$$\Pi^{pe} > 2\left(\Pi^{dca} - \Pi^{mix1}\right) \tag{A-35}$$

and for *r* if and only if

$$(1-\beta)\Pi^{pe} > \Pi^{dca} - 2\beta \left(\Pi^{dca} - \Pi^{mix1}\right)$$
(A-36)

This deviation is never profitable when $b \leq \overline{b}_{nt,ne}(a)$. To see this, rewrite (A-36) as

$$(1-\beta)\left[\Pi^{pe} - 2\left(\Pi^{dca} - \Pi^{mix1}\right)\right] > \Pi^{dca} - 2\left(\Pi^{dca} - \Pi^{mix1}\right)$$
(A-37)

When (A-35) holds the term in square brackets in the left-hand side of (A-37) is positive. The condition is therefore most likely to hold for $\beta = 0$, where it becomes $\Pi^{pe} > \Pi^{dca}$, which is therefore a necessary condition for the deviation to be profitable for r. Since $\Pi^{pe} > \Pi^{dca}$ never holds for $b \leq \overline{b}_{nt,ne}(a)$, this deviation to mutual exclusivity by $\{s, r\}$ does not further restrict the region of parameters for the existence of an equilibrium with g = dca relative to the case without exclusive contracts.

Finally, deviations to g = dm and g = um are not self enforcing since, as will be proven below, there always exist profitable and self-enforcing deviations from these networks also with exclusive contracts.

Existence of equilibria with $g_{nt,e}^* = pe$

As in Proposition 3, consider an equilibrium with g = pe supported by mutual exclusivity between *s* and *r* and between *s'* and *r'*. Consider first a deviation to g = dca by the grand coalition. This deviation is profitable for suppliers and retailers if and only if, respectively, (A-35) and (A-36) fail (or, more precisely, if the sign in those conditions is changed from > to <). With a reasoning analogous to that conducted above in relation to (A-37) one can demonstrate that the deviation is always profitable for retailers when it is profitable for suppliers, which implies that there exists a jointly profitable deviation to g = dca if and only if (A-35) fails, which is the case if and only if $b \leq \overline{b}_{nt,e}(a)$. Since $\overline{b}_{nt,e}(a) < \overline{b}_{nt,e}(a)$, the deviation to g = dca is self-enforcing whenever it is profitable. Therefore, the existence of an equilibrium with g = pe requires (A-35) to hold, i.e., $b > \overline{\overline{b}}_{nt,e}(a)$.

Consider next a deviation to g = mix by the grand coalition. This deviation is never profitable for the supplier that would be left dealing with only one retailer, since it can be verified that with linear demand $\beta (\Pi^{mix2} - \Pi^{um}) > \beta \Pi^{pe}$. Moreover, it can be verified that a deviation to g = dmis never profitable for suppliers, because $\Pi^{dm} - \Pi^{bm} \leq \Pi^{pe}$, and a deviation to g = um is never profitable for retailers, because $\Pi^{um} \leq \Pi^{pe}$. Intuitively, although a deviation to g = dm may increase total industry profits when $\Pi^{dm} > 2\Pi^{pe}$, it reduces the bargaining power of suppliers, who, after committing to a single retailer, do not have an outside option when bargaining in stage 2. A deviation to g = um always reduces industry profits (as well as the bargaining power of retailers) and is thus never profitable.

Existence of equilibria with $g_{nt,e}^* = mix$

As in Proposition 3, consider a candidate equilibrium with g = mix in which s' commits to exclusivity with r, r' commits to exclusivity with s, and s and r trade with each other on a nonexclusive basis. A deviation to g = dca requires consent by all firms and can never be profitable for both suppliers. Specifically, the deviation is profitable for s (i.e., the supplier dealing with two retailers in the candidate equilibrium) if and only if $2\left(\Pi^{dca} - \Pi^{mix1}\right) > \Pi^{mix2} - \Pi^{pe}$ and for s' (i.e., the supplier dealing with only one retailer in the candidate equilibrium) if and only if $2\left(\Pi^{dca} - \Pi^{mix1}\right) > \Pi^{mix2} - \Pi^{pe}$ and for s' (i.e., the supplier dealing with only one retailer in the candidate equilibrium) if and only if $2\left(\Pi^{dca} - \Pi^{mix1}\right) > \Pi^{mix2} - \Pi^{um}$. It can be verified that with linear demand these two conditions cannot hold in the same parameter region.

Consider now a deviation by a coalition $\{s, r\}$ in which r commits to exclusivity with s, thus implementing g = um and excluding s' is never profitable for r. To see this note that such a deviation is profitable for r if and only if $(1 - \beta) \Pi^{um} > \Pi^{mix2} - \beta (2\Pi^{mix2} - \Pi^{pe} - \Pi^{um})$, which can be re-written as $(2 - 1/\beta) (\Pi^{mix2} - \Pi^{um}) > \Pi^{pe}$. Since the left-hand side is increasing in β , this condition is most likely to hold when $\beta = 1$. When this is the case, the condition becomes $\Pi^{mix2} > \Pi^{pe} + \Pi^{um}$, which is never the case with linear demand. This implies that this condition never holds and thus that there never exists a mutually profitable deviation to g = um.

Consider next a deviation by a coalition $\{s, r\}$ in which *s* commits to exclusivity with *r*, thus implementing g = dm and excluding *r'*. This deviation is profitable for *s* if and only if $\Pi^{dm} - \Pi^{bm} > \Pi^{mix1} + \Pi^{mix2} - \Pi^{pe}$, which is never the case when (A-33) holds.

Finally, consider deviations in which *s* unilaterally rescinds his link with *r* or vice versa. Both deviations implement g = pe. The unilateral deviation by *s* is not profitable if and only if (A-33) holds and the unilateral deviation by *r* is not profitable if and only if (A-34) holds. As demonstrated above, (A-34) holds for all $\beta \in [0, 1]$ when (A-33) holds. Therefore there exist no profitable unilateral deviations to g = pe if and only if (A-33) holds.

The analysis above implies that there exists an equilibrium with g = mix if and only if (A-33) holds.

There do not exist equilibria with $g_{nt,e}^* \in \{bm, dm, um\}$

This part of the proof is analogous to that of the proof of Proposition 6. Specifically, there does not exist equilibria with g = dm because $\pi_s^{pe} > \pi_s^{dm}$ and equilibria with g = um because $\pi_r^{pe} > \pi_r^{um}$.